## UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, DC 20549

## FORM 10-Q

$\times$	QUARTERLY REPORT PURSUANT TO	SECTION 13 OR 15(d) OF THE S	ECURITIES EXCHANGE ACT OF 1934
	For the quarterly period ended	<b>September 30, 2019</b>	
	TRANSITION REPORT PURSUANT TO	SECTION 13 OR 15(d) OF THE SI	ECURITIES EXCHANGE ACT OF 1934
	For the Transition Period From		
C	' ' E' N 1 004 37350		
Com	mission File Number <u>001-36378</u>		
		PROFIRE I	ENERGY, INC.
			rant as specified in its charter)
			20.000.025
	Nevada (State or other jurisdiction of incorpor	ntion or organization)	20-0019425
	321 South 1250 West,	,	(I.R.S. Employer Identification No.)
	Lindon, Utah	Suite 1	84042
	(Address of principal execu	tive offices)	(Zip Code)
	(Address of principal execu	tive offices)	(Zip Code)
		(801	) 796-5127
		(Registrant's telephone	number, including area code)
			filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 ) and (2) has been subject to such filing requirements for the past 90 days. Yes $\boxtimes$ No $\square$
			ractive Data File required to be submitted pursuant to Rule 405 of Regulation S-T (§ nat the registrant was required to submit such files). Yes $\boxtimes$ No $\square$
			lerated filer, a non-accelerated filer, smaller reporting company, or an emerging growth reporting company," and "emerging growth company" in Rule 12b-2 of the Exchange Act.
Larg	ge accelerated filer	Acce	lerated Filer ⊠
Non	-accelerated filer □	Smal	ler reporting company ⊠
		Emer	rging growth company □
	emerging growth company, indicate by checunting standards provided pursuant to Section		not to use the extended transition period for complying with any new or revised financial
Indic	eate by check mark whether the registrant is a	a shell company (as defined in Rule	12b-2 of the Exchange Act.) Yes□ No ⊠
Secu	rities registered pursuant to Section 12(b) of	the Act:	
Title	e of each class	Trading Symbol(s)	Name of each exchange on which registered
	amon, \$0.001 Par Value	PFIE	NASDAQ
COIL	inion, 40.001 Lat Value	1 f IL	νησωίν

 strant had 50,807,831 sha	ics of common stock is	ssucu aner / ,373,433 SI	iares of confinion stock (	oustanding, par value s	0.001.

# PROFIRE ENERGY, INC. FORM 10-Q TABLE OF CONTENTS

	Page
PART I — FINANCIAL INFORMATION	
Item 1. Financial Statements	
Condensed Consolidated Balance Sheets	4
Condensed Consolidated Statements of Operations and Comprehensive Income (Unaudited)	5
Condensed Consolidated Statements of Cash Flows (Unaudited)	6
Condensed Consolidated Statements of Stockholders' Equity (Unaudited)	7
Notes to Condensed Consolidated Financial Statements (Unaudited)	9
Item 2. Management's Discussion and Analysis of Financial Condition And Results of Operations	16
Item 3. Quantitative and Qualitative Disclosure about Market Risk	20
Item 4. Controls and Procedures	20
PART II — OTHER INFORMATION	
Item 1. Legal Proceedings	21
Item 1A. Risk Factors	21
Item 2. Unregistered Sales of Equity Securities and Use of Proceeds	21
Item 3. Defaults Upon Senior Securities	21
Item 4. Mine Safety Disclosures	21
Item 5. Other Information	21
Item 6. Exhibits	22
Signatures	23

### PART I. FINANCIAL INFORMATION Item 1 Financial Information

#### PROFIRE ENERGY, INC. AND SUBSIDIARIES

Condensed Consolidated Balance Sheets

		As	of	
	Sep	tember 30, 2019	Γ	December 31, 2018
		(Unaudited)		
CURRENT ASSETS				
Cash and cash equivalents	\$	9,944,128	\$	10,101,932
Short-term investments		864,629		961,256
Short-term investments - other		2,000,000		3,596,484
Accounts receivable, net		6,568,599		6,885,296
Inventories, net (note 3)		9,782,643		9,659,571
Prepaid expenses & other current assets		1,076,138		473,726
Income tax receivable		524,695		173,124
Total Current Assets		30,760,832		31,851,389
LONG-TERM ASSETS				
Net deferred tax asset		_		85,092
Long-term investments		7,319,099		7,978,380
Financing right-of-use asset		128,738		_
Property and equipment, net		10,896,855		8,020,462
Intangible assets, net		3,934,727		429,956
Goodwill		1,120,381		997,701
Total Long-Term Assets		23,399,800		17,511,591
TOTAL ASSETS	\$	54,160,632	\$	49,362,980
CURRENT LIABILITIES				
Accounts payable	\$	2,181,592	\$	1,177,985
Accrued vacation		446,451		311,435
Accrued liabilities		2,209,303		1,445,510
Current financing lease liability (note 8)		67,984		_
Income taxes payable		627,010		1,172,191
Total Current Liabilities		5,532,340		4,107,121
LONG-TERM LIABILITIES				
Net deferred income tax liability		134,046		_
Long-term financing lease liability		63,951		_
TOTAL LIABILITIES		5,730,337		4,107,121
STOCKHOLDERS' EQUITY (note 4)				
Preferred stock: \$0.001 par value, 10,000,000 shares authorized: no shares issued or outstanding				
Common stock: \$0.001 par value, 100,000,000 shares authorized: 100 shares issued of outstanding at September 30, 2019, and 49,707,805 issued		_		_
and 47,932,305 outstanding at December 31, 2018		50,762		49,708
Treasury stock, at cost		(4,859,230)		(2,609,485)
Additional paid-in capital		29,608,685		28,027,742
Accumulated other comprehensive loss		(2,629,369)		(2,895,683)
Retained earnings		26,259,447		22,683,577
TOTAL STOCKHOLDERS' EQUITY		48,430,295		45,255,859
TOTAL LIABILITIES AND STOCKHOLDERS' EQUITY	\$	54,160,632	\$	49,362,980

Condensed Consolidated Statements of Operations and Comprehensive Income (Unaudited)

(Unaudited	<i>'</i>	For		ths E	nded September	Fo	or the Nine Mon	ths Er 30,	nded September
			2019		2018		2019		2018
REVENUES (note 6)						'			
Sales of goods, net		\$	9,251,947	\$	10,830,592	\$	29,009,837	\$	33,009,616
Sales of services, net			653,814		669,310		1,853,013		1,999,764
Total Revenues			9,905,761		11,499,902		30,862,850		35,009,380
COST OF SALES									
Cost of goods sold-product			4,326,335		4,917,449		13,465,989		15,434,698
Cost of goods sold-services			410,130		484,327		1,275,655		1,437,749
Total Cost of Goods Sold			4,736,465	_	5,401,776	_	14,741,644	_	16,872,447
GROSS PROFIT			5,169,296		6,098,126		16,121,206		18,136,933
OPERATING EXPENSES									
General and administrative expenses			3,256,023		3,180,725		9,984,251		9,887,451
Research and development			641,716		377,676		1,503,645		1,097,897
Depreciation and amortization expense			130,105		143,327		357,238		401,114
Total Operating Expenses			4,027,844		3,701,728		11,845,134		11,386,462
INCOME FROM OPERATIONS			1,141,452		2,396,398		4,276,072		6,750,471
OTHER INCOME (EXPENSE)									
Gain on sale of fixed assets			34,826		43,904		73,166		129,989
Other expense			(2,065)		(1,506)		(3,029)		(7,462)
Interest income			38,478		85,167		216,068		310,646
Total Other Income			71,239		127,565		286,205		433,173
INCOME BEFORE INCOME TAXES			1,212,691		2,523,963		4,562,277		7,183,644
INCOME TAX EXPENSE			290,943		864,874		986,407		1,934,057
NET INCOME		\$	921,748	\$	1,659,089	\$	3,575,870	\$	5,249,587
OTHER COMPREHENSIVE INCOME (LOSS)									
Foreign currency translation gain (loss)		\$	(91,397)	\$	170,641	\$	160,453	\$	(223,431)
Unrealized gains (losses) on investments		·	(12,386)		(11,963)		105,861		(35,972)
Total Other Comprehensive Income (Loss)			(103,783)	_	158,678	_	266,314		(259,403)
Total out complete and (2000)			(105,705)		150,070	_	200,511		· · · · ·
COMPREHENSIVE INCOME		\$	817,965	\$	1,817,767	\$	3,842,184	\$	4,990,184
BASIC EARNINGS PER SHARE (note 7)		\$	0.02	\$	0.03	\$	0.08	\$	0.11
FULLY DILUTED EARNINGS PER SHARE		\$	0.02	\$	0.03	\$	0.07	\$	0.11
BASIC WEIGHTED AVG NUMBER OF SHARES OUTSTANDING			47,739,192		48,082,506		47,509,357		48,337,517
FULLY DILUTED WEIGHTED AVG NUMBER OF SHARES OUTSTANDING	:		48,469,246	_	48,852,167	_	48,259,900	_	49,107,178

Condensed Consolidated Statements of Cash Flows (Unaudited)

For the Nine Months Ended September 30,

		7
	2019	2018
OPERATING ACTIVITIES		
Net income	\$ 3,575,870	\$ 5,249,587
Adjustments to reconcile net income to net cash provided by operating activities:		
Depreciation and amortization expense	732,396	667,085
Gain on sale of fixed assets	(73,166)	(120,825)
Bad debt expense	255,943	134,901
Stock awards issued for services	358,270	916,795
Changes in operating assets and liabilities:		
Changes in accounts receivable	1,244,104	(184,951)
Changes in income taxes receivable/payable	(890,523)	(432,575)
Changes in inventories	1,711,446	(3,863,287)
Changes in prepaid expenses	(586,576)	(172,497)
Changes in deferred tax asset/liability	219,138	22,564
Changes in accounts payable and accrued liabilities	855,207	1,506,396
Net Cash Provided by Operating Activities	7,402,109	3,723,193
INVESTING ACTIVITIES		
Proceeds from sale of equipment	75,310	219,269
Sale (purchase) of investments	2,476,227	(876,463)
Purchase of fixed assets	(3,309,191)	(1,271,997)
Payments for acquisitions	(4,322,722)	
Net Cash Used in Investing Activities	(5,080,376)	(1,929,191)
FINANCING ACTIVITIES		
Value of equity awards surrendered by employees for tax liability	(185,004)	(737,024)
Cash received in exercise of stock options	8,870	174,002
Purchase of treasury stock	(2,249,745)	(4,000,000)
Principal paid towards lease liability	(53,190)	
Net Cash Used in Financing Activities	(2,479,069)	(4,563,022)
Effect of exchange rate changes on cash	(468)	(38,941)
NET DECREASE IN CASH	(157,804)	(2,807,961)
CASH AT BEGINNING OF PERIOD	10,101,932	11,445,799
CASH AT END OF PERIOD	\$ 9,944,128	\$ 8,637,838
SUPPLEMENTAL DISCLOSURES OF CASH FLOW INFORMATION		
CASH PAID FOR:		
Interest	\$ 4,469	\$ —
Income taxes	\$ 1,793,281	\$ 2,164,149
NON-CASH FINANCING AND INVESTING ACTIVITIES:		
Issuance of common stock - Midflow acquisition	\$ 1,020,000	\$ —

Condensed Consolidated Statements of Stockholders' Equity (Unaudited)

	Comm	Common Stock		Accumulated Other Additional Paid-In Comprehensive						al Stockholders'				
	Shares		Amount	Ac	Capital			e (Loss)	Т	reasury Stock	Re	tained Earnings	101	Equity
Balance, December 31, 2018	47,932,305	\$	49,708	\$	28,027,742	\$	§ (	2,895,683)	\$	(2,609,485)	\$	22,683,577	\$	45,255,859
Stock based compensation					66,714	1								66,714
Stock issued in exercise of stock options	2,483		2		(2)									_
Stock issued in settlement of RSUs	148,723		149		379,712									379,861
Tax withholdings paid related to stock based compensation					(143,022)									(143,022)
Treasury stock repurchased	(775,287)									(1,333,578)				(1,333,578)
Foreign currency translation								149,415						149,415
Unrealized gains on investments								68,752						68,752
Net income												1,668,618		1,668,618
Balance, March 31, 2019	47,308,224	\$	49,859	\$	28,331,144	\$	\$ (	2,677,516)	\$	(3,943,063)	\$	24,352,195	\$	46,112,619
Stock based compensation					297,127									297,127
Stock issued in exercise of stock options	9,174		9		6,841									6,850
Stock issued in settlement of RSUs	148,794		149		(149)									_
Tax withholdings paid related to stock based compensation					(41,411)									(41,411)
Foreign currency translation								102,435						102,435
Unrealized gains on investments								49,495						49,495
Net income												985,504		985,504
Balance, June 30, 2019	47,466,192	\$	50,017	\$	28,593,552	\$	\$ (	2,525,586)	\$	(3,943,063)	\$	25,337,699	\$	47,512,619
Stock based compensation					(5,571)									(5,571)
Stock issued in exercise of stock options	4,836		5		2,015									2,020
Stock issued in settlement of RSUs	546		1		(1)									_
Stock issued in acquisition (note 9)	739,130		739		1,019,261									1,020,000
Tax withholdings paid related to stock based compensation					(571)									(571)
Treasury stock repurchased	(592,100)									(916,167)				(916,167)
Foreign currency translation								(91,397)						(91,397)
Unrealized losses on investments								(12,386)						(12,386)
Net income												921,748		921,748
Balance, September 30, 2019	\$ 47,618,604	\$	50,762	\$	29,608,685	9	\$ (	2,629,369)	\$	(4,859,230)	\$	26,259,447	\$	48,430,295

 $\label{the condensed consolidated financial statements.} The accompanying notes are an integral part of these condensed consolidated financial statements.}$ 

Condensed Consolidated Statements of Stockholders' Equity (Continued) (Unaudited)

	Commor	Stoc	k			ccumulated Other						
	Shares		Amount	A	Additional Paid-In Capital	Comprehensive Income (Loss)	7	Γreasury Stock	Re	etained Earnings	Tota	al Stockholders' Equity
Balance, December 31, 2017	48,606,425	\$	53,931	\$	27,535,469	\$ (2,200,462)	\$	(6,890,349)	\$	25,548,835	\$	44,047,424
Stock based compensation					575,235							575,235
Stock issued in exercise of stock options	81,213		81		74,161							74,242
Stock issued in settlement of RSUs	118,778		119		(119)							_
Tax withholdings related to stock based compensation					(83,600)							(83,600)
Foreign currency translation						(239,129)						(239,129)
Unrealized losses on investments						(33,235)						(33,235)
Net income										1,876,228		1,876,228
Balance, March 31, 2018	48,806,416	\$	54,131	\$	28,101,146	\$ (2,472,826)	\$	(6,890,349)	\$	27,425,063	\$	46,217,165
Stock based compensation					281,012							281,012
Stock issued in exercise of stock options	410,421		410		99,350							99,760
Stock issued in settlement of RSUs	143,540		144		(144)							_
Tax withholdings related to stock based compensation					(652,560)							(652,560)
Treasury stock repurchased	(1,277,954)							(4,000,000)				(4,000,000)
Foreign currency translation						(154,943)						(154,943)
Unrealized gains on investments						9,226						9,226
Net income										1,714,270		1,714,270
Balance, June 30, 2018	\$ 48,082,423	\$	54,685	\$	27,828,804	\$ (2,618,543)	\$	(10,890,349)	\$	29,139,333	\$	43,513,930
Stock based compensation					62,232							62,232
Stock issued in settlement of RSUs	640		1		(1)							_
Tax withholdings related to stock based compensation					(864)							(864)
Foreign currency translation						170,641						170,641
Unrealized losses on investments						(11,963)						(11,963)
Net income										1,659,089		1,659,089
Balance, September 30, 2018	\$ 48,083,063	\$	54,686	\$	27,890,171	\$ (2,459,865)	\$	(10,890,349)	\$	30,798,422	\$	45,393,065

The accompanying notes are an integral part of these condensed consolidated financial statements.  $\boldsymbol{8}$ 

Notes to the Condensed Consolidated Financial Statements (Unaudited) For the three and nine months ended September 30, 2019, and as of December 31, 2018

#### NOTE 1 - CONDENSED FINANCIAL STATEMENTS

Except where the context otherwise requires, all references herein to the "Company," "Profire," "we," "us," "our," or similar words and phrases are to Profire Energy, Inc. and its wholly owned subsidiary, taken together.

The accompanying financial statements have been prepared by the Company without audit. In the opinion of Management, all adjustments (which include only normal recurring adjustments) necessary to present fairly the financial position, results of operations, and cash flows at September 30, 2019 and for all periods presented herein have been made.

Certain information and footnote disclosures normally included in financial statements prepared in accordance with accounting principles generally accepted in the United States of America ("US GAAP") have been condensed or omitted. These condensed consolidated financial statements should be read in conjunction with the Company's audited financial statements contained in its annual report on Form 10-K for the year ended December 31, 2018 ("Form 10-K"). The results of operations for the three and nine month periods ended September 30, 2019 and 2018 are not necessarily indicative of the operating results for the full years.

#### NOTE 2 – ORGANIZATION AND SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

#### Organization and Line of Business

This Organization and Summary of Significant Accounting Policies of the Company is presented to assist in understanding the Company's condensed consolidated financial statements. The Company's accounting policies conform to accounting principles generally accepted in the United States of America ("US GAAP").

The Company provides burner- and chemical-management products and services for the oil and gas industry primarily in the US and Canadian markets.

#### Significant Accounting Policies

There have been no changes to the significant accounting policies of the Company from the information provided in Note 1 of the Notes to the Consolidated Financial Statements in the Company's most recent Form 10-K, except as discussed below.

#### Leases

In February of 2016, the Financial Accounting Standards Board ("FASB") issued Accounting Standards Update No. 2016-02 - Leases (Topic 842), which significantly amends the way companies are required to account for leases. Under the updated leasing guidance, some leases that did not have to be reported previously are now required to be presented as an asset and liability on the balance sheet. In addition, for certain leases, what was previously classified as an operating expense must now be allocated between amortization expense and interest expense. The Company adopted this update as of January 1, 2019 using the modified retrospective transition method. Prior periods have not been restated. Upon implementation, the Company recognized an initial right-of-use asset of \$132,488 and lease liability of \$132,488. Due to the simple nature of the Company's leases, no change to retained earnings was required. See Note 8 for further details.

### Recent Accounting Pronouncements

The Company has evaluated all recent accounting pronouncements and determined that the adoption of pronouncements applicable to the Company has not had or is not expected to have a material impact on the Company's financial position, results of operations or cash flows.

#### Reclassification

Certain balances in previously issued consolidated financial statements have been reclassified to be consistent with the current period presentation. The reclassification had no impact on financial position, net income, or stockholders' equity.

The accompanying notes are an integral part of these condensed consolidated financial statements.

#### NOTE 3 – INVENTORY

Inventories consisted of the following at each balance sheet date:

		\$ 457,549 \$ 76,3 10,168,955 10,474,5				
	Sept	ember 30, 2019		December 31, 2018		
Raw materials	\$	457,549	\$	76,319		
Finished goods		10,168,955		10,474,522		
Work in process		_		_		
Subtotal		10,626,504		10,550,841		
Reserve for obsolescence		(843,861)		(891,270)		
Total	\$	9,782,643	\$	9,659,571		

#### NOTE 4 - STOCKHOLDERS' EQUITY

As of September 30, 2019, and December 31, 2018, the Company held3,142,887 and 1,775,500 shares of its common stock in treasury at a total cost of \$4,859,230 and \$2,609,485, respectively.

On April 22, 2019, the Board of Directors (the "Board") of the Company approved the 2019 Executive Incentive Plan (the "EIP") for Brenton W. Hatch, the Company's President and Chief Executive Officer, Ryan W. Oviatt, the Company's Chief Financial Officer, Cameron M. Tidball, the Company's Chief Business Development Officer, Jay G. Fugal, the Company's Vice President of Operations, and Patrick D. Fisher, the Company's Vice President of Product development. The EIP provides for the potential award of bonuses to the participants based on the Company's financial performance in fiscal year 2019. If earned, the bonuses will be payable in cash and stock, and the stock portion of the bonuses is intended to constitute an award under the Company's 2014 Equity Incentive Plan, as amended (the "Plan"). In addition to the EIP, the Board also approved as a long-term incentive the grants of restricted stock unit awards to Messrs. Oviatt, Tidball, Fugal, and Fisher pursuant to the Plan (the "2019 LTIP").

#### 2019 EIP

Under the terms of the EIP, each participating executive officer has been assigned a target bonus amount for fiscal 2019. The target bonus amount for Mr. Hatch is \$41,000, the target bonus amount for Mr. Oviatt is \$90,125, the target bonus amount for Mr. Tidball is \$84,357, the target bonus for Mr. Fugal is \$41,200, and the target bonus for Mr. Fisher is \$38,750 CAD. Under no circumstance can the participants receive more than two times the target bonus amount assigned to such participant.

Participants will be eligible to receive bonuses based upon reaching or exceeding performance goals established by the Board or its Compensation Committee for fiscal 2019. The performance goals in the EIP are based on the Company's total revenue, net income, free cash flow, and product development milestones. Each of these performance goals will be weighted 25% in calculating bonus amounts.

The bonus amounts earned under the EIP, if any, will be paid 50% in cash and 50% in shares of Restricted Stock under the Plan. In no event shall the total award exceed 200% of the target bonus amount for each participant, or exceed any limitations otherwise set forth in the Plan. The actual bonus amounts, if any, will be determined by the Compensation Committee of the Board upon the completion of fiscal 2019 and paid by March 15, 2020, subject to all applicable tax withholding.

### 2019 LTIP

The 2019 LTIP consists of total awards of up to 66,213 restricted stock units ("Units") to Mr. Oviatt, up to 51,646 Units to Mr. Tidball, up to 35,313 Units to Mr. Fugal, and up to 24,862 Units to Mr. Fisher pursuant to two separate Restricted Stock Unit Award Agreements to be entered between the Company and each participant. One agreement covers 33% of each award recipient's Units that are subject to time-based vesting, and the other agreement covers the remaining 67% of such award recipient's Units that may vest based on performance metrics. Upon vesting, the award agreements entitle the award recipients to receive one share of the Company's common stock for each vested Unit. The vesting period of the 2019 LTIP began on January 1, 2019 and terminates on December 31, 2021.

On March 14, 2019, the Board approved a grant of 85,000 restricted stock units ("RSUs") to various employees. The awards vest annually overfive years and will result in a total compensation expense of \$149,600 to be recognized over the vesting period.

On June 12, 2019, the Board approved a grant of 183,942 RSUs to Independent Directors. Half of the RSUs vest immediately on the date of grant and the remaining 50% of the RSUs will vest on the first anniversary of the grant date or at the Company's next Annual Meeting of Stockholders, whichever is earlier. The awards will result in total compensation expense of \$252,000 to be recognized over the vesting period.

#### NOTE 5 - SEGMENT INFORMATION

The Company operates in the United States and Canada. Segment information for these geographic areas is as follows:

		For the Three Months	End	ed September 30,		For the Nine Months	ns Ended September 30,			
Sales	2019			2018	2019			2018		
Canada	\$	1,890,592	\$	1,803,957	\$	3,883,010	\$	4,374,844		
United States		8,015,169		9,695,945		26,979,840		30,634,536		
Total Consolidated	\$	9,905,761	\$	11,499,902	\$	30,862,850	\$	35,009,380		
		End		For the Nine Months	ns Ended September 30,					
Profit (Loss)		2019	2018			2019	2018			
Canada	\$	(242,182)	\$	24,224	\$	(1,171,424)	\$	(676,268)		
United States		1,163,930		1,634,865		4,747,294		5,925,855		
Total Consolidated	\$	921,748	\$	1,659,089	\$	3,575,870	\$	5,249,587		
						A	s of			
Long-Lived Assets						September 30, 2019		December 31, 2018		
Canada					\$	5,281,869	\$	2,509,129		
United States						18,117,931		15,002,462		
Total Consolidated					\$	23,399,800	\$	17,511,591		

#### NOTE 6 - REVENUE

#### Performance Obligations

Our performance obligations include providing product and servicing our product. We recognize product revenue performance obligations in most cases when the product is delivered to the customer. Occasionally, if we are shipping the product on a customer's account, we recognize revenue when the product has been shipped. At that point in time, the control of the product is transferred to the customer. When we perform service work, we apply the practical expedient that allows us to recognize service revenue when we have the right to invoice the customer for the work completed. We do not engage in transactions acting as an agent. The time needed to complete our performance obligations varies based on the size of the project; however, we typically satisfy our performance obligations within a few months of entering into the applicable sales contract or service contract.

#### Contract Balances

We have elected to use the practical expedient in ASC 340-40-25-4 (regarding recognition of the incremental costs of obtaining a contract) for costs related to contracts that are estimated to be completed within one year. All of our current sales contracts and service contracts are expected to be completed within one year, and as a result, we have not recognized a contract asset account. If we had chosen not to use this practical expedient, we would not expect a material difference in the contract balances. We also did not have any material contract liabilities because we typically do not receive payments in advance of recognizing revenue.

#### Disaggregation of Revenue

All revenue recognized in the income statement is considered to be revenue from contracts with customers. The table below shows revenue by category:

	For the Three Month	s Ended S	eptember 30,		ed September 30,		
	 2019		2018		2019		2018
Electronics	\$ 3,880,542	\$	4,988,293	\$	12,632,170	\$	14,208,867
Manufactured	519,990		599,951		1,456,789		2,369,461
Re-Sell	4,851,415		5,242,348		14,920,878		16,431,288
Service	653,814		669,310		1,853,013		1,999,764
Total Revenue	\$ 9,905,761	\$	11,499,902	\$	30,862,850	\$	35,009,380

#### NOTE 7 - BASIC AND DILUTED EARNINGS PER SHARE

The following table is a reconciliation of the numerator and denominators used in the earnings per share calculation:

		For	the T	hree Months	Enc	led September 30	,		
		2019					2018		
	 Income (Numerator)	Weighted Average Shares (Denominator)			Income (Numerator)		Weighted Average Shares (Denominator)		er-Share Amount
Basic EPS									
Net income available to common stockholders	\$ 921,748	47,739,192	\$	0.02	\$	1,659,089	48,082,506	\$	0.03
Effect of Dilutive Securities									
Stock options & RSUs	_	730,054				_	769,661		
Diluted EPS									
Net income available to common stockholders + assumed conversions	\$ 921,748	48,469,246	\$	0.02	\$	1,659,089	48,852,167	\$	0.03

Options to purchase 244,600 and 251,600 shares of common stock at a weighted average price of \$3.88 and \$3.89 per share were outstanding during the three months ended September 30, 2019, and 2018, but were not included in the computation of diluted EPS because the impact of these shares would be antidilutive. These options, which expire between November 2019 and May 2020, were still outstanding at September 30, 2019.

				For	the Nine Months	Ended	l September 30,		
			2019					2018	
	Inco	me (Numerator)	Weighted Average Shares (Denominator)		Per-Share Amount	Inco	ome (Numerator)	Weighted Average Shares (Denominator)	Per-Share Amount
Basic EPS									
Net income available to common stockholders	\$	3,575,870	47,509,357	\$	0.08	\$	5,249,587	48,337,517	\$ 0.11
Effect of Dilutive Securities									
Stock options & RSUs		_	750,543				_	769,661	
Diluted EPS									
Net income available to common stockholders + assumed conversions	\$	3,575,870	48,259,900	\$	0.07	\$	5,249,587	49,107,178	\$ 0.11

Options to purchase 244,600 and 251,600 shares of common stock at a weighted average price of \$3.88 and \$3.89 per share were outstanding during the nine months ended September 30, 2019 and 2018, respectively, but were not included in the computation of diluted EPS because the impact of these shares would be antidilutive. These options, which expire between November 2019 and May 2020, were still outstanding at September 30, 2019.

#### NOTE 8 - LEASES

We have leases for office equipment and office space. The leases for office equipment are classified as financing leases and the typical term is 36 months. We have the option to extend most office equipment leases, but we do not intend to do so. Accordingly, no extensions have been recognized in the right-of-use asset or lease liability. The office equipment lease payments are not variable and the lease agreements do not include any non-lease components, residual value guarantees, or restrictions. There are no interest rates implicit in the office equipment lease agreements, so we have used our incremental borrowing rate to determine the discount rate to be applied to our financing leases. The weighted average discount rate applied to our financing leases is 4.50% and the weighted average remaining lease term is 25.3 months.

The following table shows the components of financing lease cost:

Financing Lease Cost	For the Three M September 30,		For the Nine September 3	Months Ended 0, 2019
Amortization of right-of-use assets	\$	20,746	\$	56,387
Interest on lease liabilities		1,627		4,459
Total financing lease cost	\$	22,373	\$	60,846

The following table reconciles future minimum lease payments to the discounted finance lease liability:

Years ending December 31,	Amount
2019 - remaining	\$ 21,847
2020	62,995
2021	40,921
2022	12,803
2023	_
Thereafter	
Total future minimum lease payments	\$ 138,566
Less: Amount representing interest	6,631
Present value of future payments	\$ 131,935
Current portion	\$ 67,984
Long-term portion	\$ 63,951

Because our office space leases are short-term, we have elected not to recognize them on our balance sheet under the short-term recognition exemption. During the three and nine months ended September 30, 2019, we recognized \$18,779 and \$41,097, respectively in short-term lease costs associated with office space leases.

### NOTE 9 – ACQUISITIONS

#### Millstream Energy Products

On June 18, 2019, our wholly-owned subsidiary, Profire Combustion, Inc., acquired substantially all the assets from Millstream Energy Products, LTD., a Canadian corporation ("MEP"). MEP is a privately-held Canadian company that develops a line of high-performance burners, economy burners, flame arrestor housings, secondary air control plates, and other related combustion components. MEP's full line of products became available for sale by Profire's existing sales team

immediately after closing of the transaction. These products complement our burner-management system (BMS) product offerings and should enable us to supply a larger portion of the total BMS package sale to our customers.

The acquisition was accounted for as a business combination in accordance with ASC 805, Business Combinations. The purchase accounting process has not been completed primarily because the valuation of acquired assets has not been finalized. We expect to complete the purchase accounting as soon as practicable but no later than one year from the acquisition date. We do not believe there will be material adjustments. The purchase price of \$2,325,846 USD was funded through existing cash. Of this cash purchase amount \$246,322 USD has been held back for 6 months pending satisfaction of seller obligations under the purchase agreement. MEP will also receive a 4.5% royalty on proprietary MEP product revenue generated during the next five years. Based on the estimated fair value at the time of purchase, the Company recorded an estimate of intangible assets in the amount of \$885,000 USD and estimated goodwill of \$122,680 USD pending completion of the final valuation analysis.

The purchase price calculation is a follows:

The purchase price calculation is a follows.	
Cash	\$ 2,079,524
Liabilities	246,322
	\$ 2,325,846
The following table summarizes the preliminary estimated fair value of the assets acquired and liabilities assumed at the date of purchase:	
Accounts receivable	\$ 308,431
Inventory	1,123,922
Intangibles assets	885,000
Goodwill	122,680
Accounts payable	(114,187)
	\$ 2,325,846

Transaction and related costs directly related to the acquisition of MEP, consisting primarily of professional fees and integration expenses, have amounted to approximately \$66,430, were expensed as incurred and are included in general and administrative expenses.

#### Midflow Services

On August 5, 2019, we acquired all of the outstanding membership interests of Midflow Services, LLC ("Midflow"). Midflow is based in Millersburg, Ohio. Midflow provides packaged combustion solutions and services to the upstream and midstream oil and gas industry.

The acquisition was accounted for as a business combination in accordance with ASC 805, *Business Combinations*. The purchase accounting process has not been completed primarily because the valuation of the acquired assets has not been finalized. We expect to complete the purchase accounting as soon as practicable but no later than one year from the acquisition date. We do not believe there will be material adjustments. The purchase price of \$3,419,617 was funded through a combination of existing cash and shares of the Company's common stock. The cash portion of the purchase price includes \$500,000 placed in an escrow account for 12 months pending satisfaction of certain obligations under the purchase agreement. Based on the estimated fair value at the time of purchase, the Company recorded an estimate of intangible assets in the amount of \$2,648,808 which may include goodwill once the final valuation analysis is completed.

The purchase price calculation is as follows:	
Cash	\$ 2,399,617
Stock	1,020,000
	\$ 3,419,617
The following table summarizes the preliminary estimated fair value of the assets acquired and liabilities assumed at the date of purchase:	
Cash	\$ 172,850
Accounts receivable	324,989
Inventory	264,986
Prepaid expenses	14,878
Property and equipment	134,500
Intangibles and goodwill	2,648,808

Transaction costs directly related to the acquisition of Midflow, consisting primarily of professional fees and integration expenses, amounted to approximately \$44,087. All of these costs were expensed as incurred and are included in general and administrative expenses.

(134,956)

(6,438) 3,419,617

## NOTE 10 – SUBSEQUENT EVENTS

Accounts payable

Accrual liabilities

In accordance with ASC 855 "Subsequent Events," Company Management reviewed all material events through the date this report was issued and there were no subsequent events to report.

#### Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

This discussion summarizes the significant factors affecting our consolidated operating results, financial condition, liquidity, and capital resources during the three and ninemonth periods ended September 30, 2019 and 2018. This Management's Discussion and Analysis of Financial Condition and Results of Operations should be read in conjunction with the *Financial Statements* and *Notes to the Financial Statements* contained in this quarterly report on Form 10-Q and our annual report on Form 10-K for the year ended December 31, 2018.

#### Forward-Looking Statements

This quarterly report on Form 10-Q contains forward-looking statements within the meaning of Section 27A of the Securities Act of 1933, as amended (the "Exchange Act"), and Section 21E of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), that are based on Management's beliefs and assumptions and on information currently available to Management. For this purpose, any statement contained in this report that is not a statement of historical fact may be deemed to be forward-looking, including, but not limited to, statements relating to our future actions, intentions, plans, strategies, objectives, results of operations, cash flows and the adequacy of or need to seek additional capital resources and liquidity. Words such as "may," "should," "expect," "project," "plan," "anticipate," "believe," "estimate," "intend," "budget," "forecast," "predict," "potential," "continue," "should," "could," "will," or comparable terminology or the negative of such terms are intended to identify forward-looking statements; however, the absence of these words does not necessarily mean that a statement is not forward-looking. Forward-looking statements by their nature involve known and unknown risks and uncertainties and other factors that may cause actual results and outcomes to differ materially depending on a variety of factors, many of which are not within our control. Such factors include, but are not limited to, economic conditions generally and in the oil and gas industry in which we and our customers participate; competition within our industry; legislative requirements or changes which could render our products or services less competitive or obsolete; our failure to successfully develop new products and/or services or to anticipate current or prospective customers' needs; price increases; limits to employee capabilities; delays, reductions, or cancellations of contracts we have previously entered into; sufficiency of working capital, capital resources and liquidity and other factors detailed herein and in

Forward-looking statements are based on current industry, financial and economic information which we have assessed but which by its nature is dynamic and subject to rapid and possibly abrupt changes. Due to risks and uncertainties associated with our business, our actual results could differ materially from those stated or implied by such forward-looking statements. Moreover, neither we nor any other person assumes responsibility for the accuracy and completeness of these forward-looking statements and we hereby qualify all of our forward-looking statements by these cautionary statements.

Forward-looking statements in this report are based only on information currently available to us and speak only as of the date on which they are made. We undertake no obligation to amend this report or revise publicly these forward-looking statements (other than as required by law) to reflect subsequent events or circumstances, whether as the result of new information, future events or otherwise.

The following discussion should be read in conjunction with our financial statements and the related notes contained elsewhere in this report and in our other filings with the Commission.

#### Overview

We are an oilfield technology company providing products that enhance the efficiency, safety, and compliance of the oil and gas industry. We specialize in the design of burner and combustion management solutions utilized on heated applicances throughout the oil and gas industry. We sell our products and services primarily throughout North America. Our experienced team of industry service professionals also provides supporting services for our products.

#### **Principal Products and Services**

Across the energy industry, there are numerous demands for heat generation and control. Applications such as combustors, enclosed flares, gas production units, treaters, glycol and amine reboilers, indirect line-heaters, heated tanks, and process heaters require heat as part of their production or processing functions. This is provided by a burner flame. This burner flame is integral to the process of separating, treating, storing, and transporting oil and gas. Factors such as the gravity, the presence of hydrates, temperature and hydrogen sulfide content contribute to the requirement for heat in oil and gas production and processing applications. Our burner-management systems ignite, monitor, and manage this burner flame, which can be operated remotely, reducing the need for employee interaction with the burner, such as for the purposes of re-ignition or temperature monitoring. In addition, our burner-management systems can help reduce gas emissions by quickly reigniting a failed flame.

Oil and gas producers utilize burner-management systems to achieve increased safety, greater operational efficiencies, and improved compliance with changing industry regulations. Without burner-management systems, an employee must discover and reignite an extinguished burner flame, then restart the application manually. Therefore, without burner-management systems, all application monitoring is done directly on-site. Such on-site monitoring can result in the interruption of production for longer periods of time, risk of reigniting a flame, which can lead to burns and explosions, and the possibility of raw gas being vented into the atmosphere when the flame fails. In addition, without a burner-management system, burners often run longer, incurring significant fuel costs. We believe there is a growing trend in the oil and gas industry toward enhanced control, process automation, and data logging, largely for improved efficiency and operational cost savings, and partly for potential regulatory-satisfaction purposes. Our burner-management systems are designed to be always on standby to make sure the burner flame is lit and managed properly, which can reduce how often a burner is running and may reduce fuel costs. We continue to assess compliance-interest in the industry, and we believe that enhanced burner-management products and services can help our customers be compliant with such regulatory requirements, where applicable. In addition to selling products, we train and dispatch service technicians to service burner flame installations throughout the United States and Canada.

We initially developed our first burner-management system in 2005. Since then, we have released several iterations of our initial burner-management system, increasing features and capabilities, while maintaining compliance with North American standards including, Canadian Standards Association (CSA), Underwriters Laboratories (UL), and Safety Integrity Level (SIL) standards.

Our burner-management systems have become widely used in Western Canada and throughout many regions in the United States. We have sold our burner-management systems to many large energy companies, including Anadarko, Chesapeake, ConocoPhillips, Devon, Encana, XTO, CNRL, Shell, OXY, and others. Our systems have also been sold or installed in other parts of the world, including many countries in Europe, South America, Africa, the Middle East and Asia. We are established in the North American oil and gas markets, which is our current primary focus, but we are working to expand further into other international markets.

#### Product Extensions:

The PF3100 is an advanced burner and combustion management system which is designed to operate, monitor, control, and manage a wide variety of complex, heated appliances. Throughout the industry, Programmable Logic Controllers, or PLCs, are used to operate and manage custom-built oilfield applications. Though capable, PLCs can be expensive, difficult to install and maintain. The PF3100 can help oilfield producers meet deadlines and improve profitability through an off-the-shelf solution with dynamic customization. We are selling the PF3100 for initial use in the oil and gas industry's natural-draft and forced-draft applications.

We recognized the area of burner management most in need of innovation was the user experience. The PF2200 is designed to optimize installation, commissioning, troubleshooting and daily operation. This focus on the user will optimized the time required on-site for both installation and operator training. With the user focused design being combined with an expanded feature set, the PF2200 becomes a very powerful tool to reduce downtime and lessen the burden on producers in a wide variety of applications, ranging from dual-burner to forced draft, to a variety of waste-gas destruction applications.

We frequently assess market needs by participating in industry conferences and soliciting feedback from existing and potential customers, allowing us to provide quality solutions to the oil- and gas-producing companies we serve. Upon

identifying a potential market need, we begin researching the market and developing products that are likely to have feasibility for future sale.

#### Additional Complementary Products

In addition to our burner- and combustion-management systems, we also supply complementary products that provide our customers with a complete solution. These products include safety and monitoring devices such as shut-down and temperature valves, pressure transmitters and switches, burners, pilots, and other combustion related equipment. We continue to invest in the development of innovative, complementary, products which we anticipate will help bolster continued long-term growth.

#### Chemical-Management Systems

In addition to the burner-management systems and complementary technologies we sell, we also offer chemical-management systems.

Chemical injection is used for a wide variety of purposes in the oil and gas industry including down-hole inhibition of wax, hydrates, and corrosion agents, so that product can flow more efficiently to the wellhead. Once at the wellhead, chemical injection can also be used to further process the oil or gas before it is sent into a pipeline, and with other applications.

Currently, a variety of pumps are used to meter the chemicals injected but are often inaccurate in injecting the proper volume of chemical, as they may not account for all of the variables that affect how much chemical should be injected (e.g., pressure, hydrogen sulfide concentration, etc.) nor the optimal efficiency rates of varying pump systems.

Inaccurate injection levels are problematic because the chemicals injected are expensive, and over-injection causes unnecessary expense for producers. Under-injection can also be problematic because it often results in the creation of poor product (i.e., with wax, hydrate, or corrosion agents) and causes problems with pipeline operations.

Our chemical-management systems monitor and manage the chemical-injection process to ensure that optimal levels of chemicals are injected. This improves the efficiency of the pump and production quality of the well, improves safety for workers that would otherwise be exposed to these chemicals, and improves compliance with pipeline operators. Like our burner-management systems, our chemical-management systems can be monitored and managed remotely via supervisory control and data acquisition or other remote-communication systems. We hold a U.S. patent related to our chemical management system and its process for supplying a chemical agent to a process fluid.

#### **Results of Operations**

#### Comparison quarter over quarter

The table below presents certain financial data comparing the most recent quarter to prior quarters:

		For the three months ended								
	Sep	tember 30, 2019		June 30, 2019		March 31, 2019	1	December 31, 2018		September 30, 2018
Total Revenues	\$	9,905,761	\$	10,124,031	\$	10,833,058	\$	10,605,155	\$	11,499,902
Gross Profit Percentage		52.2 %	)	51.2 %	,	53.2 %	,	44.9 %		53.0 %
Operating Expenses	\$	4,027,844	\$	4,190,479	\$	3,626,811	\$	3,541,209	\$	3,701,728
Net Income	\$	921,748	\$	985,504	\$	1,668,618	\$	831,404	\$	1,659,089
Operating Cash Flow	\$	2,094,454	\$	2,699,154	\$	2,608,501	\$	1,829,363	\$	599,862

Revenues for the quarter ended September 30, 2019 decreased by 14% or \$1,594,141 compared to the quarter ended September 30, 2018, which was mostly driven by macro industry changes over that same period. The average oil price in the first nine months of 2019 was only \$57.04 per barrel compared to \$66.93 per barrel in the first nine months of 2018 representing a decrease of 14.8%. The Q3 2019 weekly average rig count for North America was 1,024 compared to 1,239 in the same period of last year. These decreases have followed the decline in oil prices in Q4 2018. As a result of these macro trends, we believe many exploration and production companies have pulled back on capital expenditure budgets or deferred

planned spending. Some customers have indicated to us that they intend to use cash in the near future to fund further debt reductions, increasing dividend payments or to initiate or expand stock repurchase programs.

Our gross profit margin was down slightly from the same quarter of last year but remains within our normally exepected range for a completed quarter. The gross margin percentage fluctuates each quarter due to changes in product mix and product related reserves. Over the past year gross profit has stayed fairly consistent, within an expected range, except for Q4 2018 when gross profit decreased due to inventory adjustments related to our chemical-management system product line. We anticipate gross margin will stay within the expected range going forward, assuming normal product mix fluctuations.

Operating expenses increased \$326,116 from the same quarter of last year, which was consistent with our previously announced strategic investment plan for 2019. As reported in our Annual Report on Form 10-K for the year ended December 31, 2018, we initiated investments during 2019 in the areas of research and development efforts in order to expand our product offerings, international market evaluation and expansion and an expanded sales and labor force to meet expected market demand and support our expanding product offerings. We intend to continue to fund these activities during the fourth quarter.

Due to the lower revenues and higher operating cost to support strategic initiatives discussed above, net income decreased 44% during the quarter ended September 30, 2019 compared to the same quarter in 2018. Operating cash flows increased significantly during the third quarter of 2019 compared to the third quarter of 2018, despite lower revenue and increased costs. This increase is primarily due to favorable movements in net working capital during the third quarter of 2019 compared to unfavorable movements in net working capital in the third quarter of 2018.

Despite the ongoing volatility in the oil and gas industry and macro trends in the short term, we remain focused on the Company's future. Assuming we successfully implement our planned investments in research and development, in international market expansion and an expanded sales and labor force, we believe we are well-positioned for continued growth in future periods, particularly if oil prices improve and allow our customers to make higher levels of capital expenditures.

#### Comparison of the nine-months ended September 30, 2019 and 2018

The table below presents certain financial data comparing the nine months ended September 30, 2019 to the same period ended September 30, 2018:

	F	or the Nine Month	is Ende	d September 30,		
		2019		2018	 \$ Change	% Change
Total Revenues	\$	30,862,850	\$	35,009,380	\$ (4,146,530)	(11.8)%
Gross Profit Percentage		52.2 %	· )	51.8 %		0.4 %
Operating Expenses	\$	11,845,134	\$	11,386,462	\$ 458,672	4.0 %
Net Income	\$	3,575,870	\$	5,249,587	\$ (1,673,717)	(31.9)%
Operating Cash Flow	\$	7,402,109	\$	3,723,193	\$ 3,678,916	98.8 %

Revenues during the nine-month period ended September 30, 2019 compared to the same period last year declined 11.8% which is largely due to a 14.8% drop in the average oil price over the same time frame, which caused our customers to reduce capital expenditure. Operating expenses increased 4.0% year-over-year as a result of ongoing strategic investments during the period. As a result of the revenue and operating cost changes, there was a 31.9% decline in net income. Our gross profit percentage increased slightly by 0.4% during the nine-months ended September 30, 2019, compared to the same period in 2018, primarily due to changes in product mix, direct labor costs, and inventory adjustments.

#### **Liquidity and Capital Resources**

Working capital at September 30, 2019 was \$25,228,492, compared to \$27,744,268 at December 31, 2018. This decrease resulted from a combination of decreases in accounts receivable and inventory which were partially offset by decreases in accrued liabilities such as income taxes payable.

We acquired land for a new office building and research and development facility in Canada in June of 2018 and construction of the facility is ongoing. Excluding the cost of the land, the total cost of the building is expected to be approximately \$4,600,000 USD. As of September 30, 2019, we had spent approximately \$3,165,000 USD towards construction of the building. We believe our available cash resources are sufficient to cover construction costs for the building and other expected capital expenditures for the foreseeable future, and we have no current plans to incur debt financing.

#### **Off-Balance Sheet Arrangements**

We have not engaged in any off-balance sheet arrangements, nor do we plan to engage in any in the foreseeable future.

#### Item 3. Quantitative and Qualitative Disclosure about Market Risk

This section is not required.

#### Item 4. Controls and Procedures

#### Evaluation of Disclosure Controls and Procedures

Our Management, with the participation of the Principal Executive Officer and Principal Financial Officer, evaluated the effectiveness of the design and operation of our disclosure controls and procedures pursuant to Rule 13a-15(b) under the Exchange Act, as of the end of the period covered by this Report. Our disclosure controls and procedures are designed to ensure that the information required to be disclosed by us in reports that we file under the Exchange Act is accumulated and communicated to our Management, including our Principal Executive Officer and Principal Financial Officer, as appropriate, to allow timely decisions regarding required disclosure and is recorded, processed, summarized and reported within the time periods specified in the rules and forms of the SEC. Based on the evaluation performed, our Management, including the Principal Executive Officer and Principal Financial Officer, concluded that the disclosure controls and procedures were effective as of September 30, 2019.

#### Changes in Internal Control over Financial Reporting

Our Management, with the participation of our Chief Executive Officer and Chief Financial Officer, evaluated the changes in our internal control over financial reporting that occurred during the quarterly period covered by this Quarterly Report on Form 10-Q. Based on that evaluation, Management concluded that no change in our internal control over financial reporting (as such term is defined in Rules 13a-15(f) and 15d-15(f) under the Exchange Act) occurred during the quarter ended September 30, 2019 that materially affected, or is reasonably likely to materially affect, our internal control over financial reporting.

#### PART II - OTHER INFORMATION

#### Item 1. Legal Proceedings

To the best of our knowledge, there are no legal proceedings pending or threatened against us that may have a material impact on us and there are no actions pending or threatened against any of our directors or officers that are adverse to us.

#### Item 1A. Risk Factors

In addition to the other information set forth in this quarterly report on Form 10-Q, you should carefully consider the risks discussed in our Annual Report on Form 10-K for the year ended December 31, 2018, which risks could materially affect our business, financial condition or future results. These risks are not the only risks facing our Company. Additional risks and uncertainties not currently known to us or that we currently deem to be immaterial may also have a material, adverse effect on our business, financial condition or future results.

#### Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

On November 5, 2018, the Company announced that the Board had authorized a share repurchase program allowing the Company, at Management's discretion, to repurchase up to \$2,000,000 worth of the Company's common stock from time to time through October 31, 2019. The repurchase program expired on October 31, 2019 and the Company does not have an active repurchase program as of the date of this report.

On August 1, 2019 the Board approved Management to purchase up to an additional \$2,000,000 worth of the Company's Common stock pursuant to the Company's existing share repurchase program.

The table below sets forth additional information regarding our share repurchases during the three months ended September 30, 2019:

Period	(a) Total Number of Shares Purchased	(b) Weighte Paid Per Sh	ed Average Price	(c) Total Number of Shares Purchased as Part of Publicly Announced Plans	Shar	Maximum Dollar Value of tes that May Yet Be hased Under the Plans
July	_	\$		_	\$	_
August	308,915	\$	1.42	308,915	\$	1,561,626
September	283,185	\$	1.69	283,185	\$	1,083,833
Total	592,100			592,100		

### Item 3. Defaults Upon Senior Securities

We do not have any debt nor any current plans to obtain debt financing.

#### **Item 4. Mine Safety Disclosures**

This item is not applicable.

#### Item 5. Other Information

This item is not applicable.

## Item 6. Exhibits

Exhibits. The following exhibits are included as part of this report:

Exhibit 10.1	Membership Interest Purchase Agreement between Profire Energy, Inc. and Midflow Services, LLC dated August 5, 2019*
Exhibit 31.1*	Certification of Principal Executive Officer Pursuant to Rule 13a-14(a)
Exhibit 31.2*	Certification of Principal Financial Officer Pursuant to Rule 13a-14(a)
Exhibit 32.1*	Certification of Brenton W. Hatch, Principal Executive Officer pursuant to 18 U.S.C. Section 1350
Exhibit 32.2*	Certification of Ryan W. Oviatt, Principal Financial Officer pursuant to 18 U.S.C. Section 1350
Exhibit 101.INS*	XBRL Instance Document
Exhibit 101.SCH*	XBRL Taxonomy Extension Schema Document
Exhibit 101.CAL*	XBRL Taxonomy Extension Calculation Linkbase Document
Exhibit 101.DEF*	XBRL Taxonomy Definition Linkbase Document
Exhibit 101.LAB*	XBRL Taxonomy Extension Label Linkbase Document
Exhibit 101.PRE*	XBRL Taxonomy Extension Presentation Linkbase Document

<sup>\*</sup> Filed herewith

## SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

PROFIRE ENERGY, INC.

Date: November 6, 2019 By: <u>/s/ Brenton W. Hatch</u>

Brenton W. Hatch Chief Executive Officer

Date: November 6, 2019 By: /s/ Ryan W. Oviatt

Ryan W. Oviatt Chief Financial Officer

## MEMBERSHIP INTEREST PURCHASE AGREEMENT by and among

PROFIRE ENERGY, INC. as Buyer,

and

DUSTIN BAKER AND BRANT BAKER, as Sellers,

with respect to all of the membership interests of

MIDFLOW SERVICES, LLC

Dated as of August 5, 2019

## TABLE OF CONTENTS

		rage
ARTICLE I	CERTAIN DEFINITIONSCertain Definitions	
ARTICLE II 2.1 2.2 2.3 2.4	PURCHASE AND SALE OF MEMBERSHIP INTERESTS	10 11 11
ARTICLE II 3.1 3.2	I CLOSING DELIVERABLES Closing Deliverables of Sellers Closing Deliveries of Buyer	14
4.1 4.2 4.3 4.4 4.5 4.6	REPRESENTATIONS AND WARRANTIES OF SELLERS	16 17 17 17 19
COMI 5.1 5.2 5.3 5.4 5.5 5.6 5.7 5.8 5.9 5.10 5.11 5.12 5.13 5.14	REPRESENTATIONS AND WARRANTIES REGARDING TRANY Organization and Power	
5.15 5.16 5.17 5.18 5.19 5.20	Litigation Brokerage Employees Product and Service Warranties Employee Benefit Plans Insurance	32 32 34

5.21	Compliance with Laws; Permits	36
5.22	Environmental Matters	37
5.23	Customers	38
5.24	Affiliate Transactions	39
5.25	Bank Accounts; Powers of Attorney	39
5.26	Disclosures	39
ARTICLE V	REPRESENTATIONS AND WARRANTIES OF BUYER	
6.1	Organization and Corporate Power	
6.2	Authorization	
6.3	Governmental Authorities and Consents	
6.4	Brokerage	
6.5	Litigation	
6.6	Acquisition for Investment	
6.7	Valid Issuance	40
ARTICLE V	II COVENANTS	
7.1	Expenses	40
7.2	Further Transfers; Transition Assistance	
7.3	Confidentiality	41
7.4	Release and Waiver	41
7.5	Non-Competition and Non-Solicitation	
7.6	Public Announcements	44
7.7	Lock-Up	44
ARTICLE V	III TAX MATTERS	45
8.1	Tax Indemnification	45
8.2	Straddle Period	45
8.3	Responsibility for Filing Pre-Closing Tax Returns	45
8.4	Responsibility for Filing Straddle Period Tax Returns	
8.5	Transfer Taxes	46
8.6	Cooperation on Tax Matters	46
8.7	Tax Sharing Agreements	46
8.8	Withholding	
ARTICLE IX	KINDEMNIFICATION	
9.1	Survival	
9.2	Indemnification	
9.3	Materiality	
9.4	Indemnification Procedures.	49
9.5	Waiver	50
9.6	Limitations on Indemnity	50
9.7	Exclusive Remedy	
ARTICLE X	MISCELLANEOUS	
10.1	Amendment and Waiver	51
10.2	Notices	52

10.3	Assignment	53
10.4	Severability	53
10.5	Interpretation	54
10.6	Entire Agreement	54
10.7	Counterparts	54
10.8	Governing Law	54
10.9	Venue	54
10.10	No Strict Construction	55
10.11	No Third-Party Beneficiaries	55
10.12	Seller Representative	55
10.13	Disclosure Schedule	56

## MEMBERSHIP INTEREST PURCHASE AGREEMENT

THIS MEMBERSHIP INTEREST PURCHASE AGREEMENT (this "Agreement") is made and entered into as of August 5, 2019 (the "Effective Date"), by and among: (i) Profire Energy, Inc., a Nevada corporation ("Buyer"); (ii) Dustin Baker and Brant Baker (each, individually, a "Seller" and, collectively, "Sellers"). Buyer and Sellers are sometimes referred to in this Agreement, individually, as a "Party" and, collectively, as the "Parties."

## Recitals:

- A. Sellers are the only members of Midflow Services, LLC, an Ohio limited liability company (the "Company").
- B. Upon the terms and subject to the conditions set forth herein, Buyer desires to acquire all of the issued and outstanding units and membership interests of the Company, including all membership, management, voting and economic interests attendant thereto (the "Membership Interests"), from Sellers, and Sellers desire to transfer such Membership Interests to Buyer; and

## Agreement:

In consideration of the foregoing and the respective representations, mutual covenants, agreements and understandings contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties agree as follows:

### ARTICLEI

## CERTAIN DEFINITIONS

- 1.1 <u>Certain Definitions</u>. For purposes of this Agreement, the following terms have the meanings set forth below:
  - "Accounting Firm" has the meaning set forth in Section 2.3(b)(v).
  - "Adjustment Amount" has the meaning set forth in Section 2.3(b)(ii).
  - "Adjustment Statement" has the meaning set forth in Section 2.3(b)(iii).
  - "Affiliate" of any particular Person means any other Person controlling, controlled by or under common control with such Person. For purposes of this definition, "control" (including the terms "controlling," "controlled by" and "under common control with") means the power to direct, or cause the direction of, the acts, management and policies of a Person, directly or indirectly, whether through the ownership of voting securities, by contract or otherwise, and such "control" will be deemed to exist if any Person directly or indirectly owns 10% or more of the voting capital stock or other ownership interests of any other Person.

- "Affiliated Group" means any affiliated group within the meaning of Section 1504(a) of the Code or any similar group defined under a similar provision of Law.
- "Applicable Rate" means the prime rate quoted from time to time by The Wall Street Journal, plus two percent (2%).
- "Balance Sheet Date" means December 31, 2018.
- "Brant Offer Letter" has the meaning set forth in Section 3.1(j).
- "Business Day" means any day other than a Saturday, a Sunday or a day on which commercial banks are closed for business in Salt Lake City, Utah.
- "Buyer" has the meaning set forth in the introductory paragraph to this Agreement.
- "Buyer Indemnified Parties" means Buyer and its Affiliates (including, following the Closing, the Company) and their respective owners, shareholders, members, partners, officers, directors, managers, employees, agents, representatives, successors and assigns.
- "Buyer Stock" means the common stock of Buyer, \$0.001 par value per share.
- "Buyer Stock Closing Price" means the price per share of Buyer Stock traded on The NASDAQ Stock Market as of the closing of trading on the last trading day of The NASDAQ Stock Market immediately preceding the Effective Date, as calculated and reported by The NASDAQ Stock Market.
- "Closing" has the meaning set forth in Section 2.4.
- "Closing Cash Amount" means the Estimated Closing Cash Amount, plus any positive Adjustment Amount (if applicable), less any negative Adjustment Amount (if applicable).
- "Closing Consideration" means the Closing Cash Amount, the Closing Stock Consideration and the Escrow Amount.
- "Closing Current Assets" means an amount, determined as of the Closing, equal to the sum of the following current assets of the Company: (a) cash and cash equivalents; (b) all Company Inventory; (c) accounts receivable, but excluding receivables from any of the Company's Affiliates, directors, employees, officers or shareholders and any of their respective Affiliates; and (d) prepaid expenses of the Company, but only to the extent that the Company actually receives the benefit or refund of such prepaid expenses following the Closing and prior to the date of the Adjustment Statement. The calculation of Closing Current Assets will be determined (y) in accordance with GAAP and (z) without giving effect to the transactions contemplated by this Agreement.
- "Closing Current Liabilities" means an amount, determined as of the Closing, equal to the sum of all of the current Liabilities of the Company, including: (a) accounts payable; (b) accrued expenses; (c) accrued Taxes; and (d) the current portion of long-term Liabilities. The calculation of Closing Current Liabilities will be determined (y) in

accordance with GAAP and (z) without giving effect to the transactions contemplated by this Agreement.

- "Closing Indebtedness" means the Indebtedness of the Company as of the Closing.
- "Closing Net Working Capital" means an amount, determined as of the Closing, equal to: (a) the Closing Current Assets, <u>less</u> (b) the Closing Current Liabilities. For purposes of calculating Closing Net Working Capital, Closing Current Assets and Closing Current Liabilities will each be determined: (x) without duplication; (y) in accordance with GAAP; and (z) without giving effect to the transactions contemplated by this Agreement.
- "Closing Stock Consideration" means that number of shares of unregistered, restricted Buyer Stock equal to: \$1,020,000, <u>divided</u> by the Buyer Stock Closing Price.
- "Code" means the Internal Revenue Code of 1986, as amended, and any reference to any particular Code section shall be interpreted to include any revision of or successor to that section regardless of how numbered or classified.
- "Company Employees" has the meaning set forth in Section 5.17(a).
- "Company Marks" means all trademarks, trade names, service marks and trade dress of owned by the Company or used in the business of the Company, including any and all rights in the names "Midflow Services, LLC," "Midflow Services," and any name or mark similar thereto, together with all goodwill associated therewith, all translations, adaptations, derivations and combinations of the foregoing, and all logos related to the foregoing.
- "Company Plan" has the meaning set forth in Section 5.19(a).
- "Confidential Information" means all information of a confidential or proprietary nature (whether or not specifically labeled or identified as "confidential" or "proprietary"), in any form or medium, whether evidenced in writing, electronic data or otherwise, that relates to the business, products, services and/or research and/or development of the Company and/or its respective suppliers, distributors, customers, independent contractors and/or other business relations. Confidential Information includes, but is not limited to, the following: (a) internal business information (including historical and projected financial information and budgets and information relating to strategic and staffing plans and practices, business, training, marketing, promotional and sales plans and practices, cost, rate and pricing structures and accounting and business methods); (b) identities of, individual requirements of, specific contractual arrangements with, and information about, suppliers, distributors, customers, independent contractors or other business relations and their confidential information; (c) trade secrets, know-how, compilations of data and analyses, techniques, systems, formulae, recipes, research, records, reports, manuals, documentation, models, data and data bases relating thereto; (d) inventions, innovations, improvements, developments, methods, designs, analyses, drawings, reports and all similar or related information (whether or not patentable); and (e) other Proprietary Rights.

- "Contract" means any written or oral contract, agreement, commitment, franchise, indenture, note, lease, deed, mortgage, purchase order, license, arrangement, understanding or course of dealing.
- "<u>Disclosure Schedule</u>" means the disclosure schedule delivered by Sellers to Buyer concurrently with the execution and delivery of this Agreement, referred to in this Agreement and dated as of the date of this Agreement.
- "Dustin Offer Letter" has the meaning set forth in Section 3.1(i).
- "Environment" means the components of the earth and includes ambient air, land, surface and sub-surface strata, groundwater, surface water, all layers of the atmosphere, all organic and inorganic matter and living organisms, and the interacting natural systems that include such components.
- "Environmental and Safety Laws" means all Laws and contractual obligations concerning public health and safety, worker health and safety, and pollution or protection of the Environment, including all those relating to the presence, use, production, generation, handling, transportation, treatment, storage, disposal, distribution, labeling, testing, processing, discharge, release, threatened release, reporting of, Remediation of, or exposure to, any Hazardous Materials, now or hereafter in effect.
- "Environmental Liabilities" means all present and future Liability relating to or arising directly or indirectly out of past, present or future Environmental damage, contamination or Remediation which arises from, relates to or was caused by the operation of the Company's business, the ownership of the Membership Interests or the ownership, possession or control of the Company's assets and properties prior to Closing, however and by whomsoever caused, and whether caused by non-compliance with, violation of or Liability under Environmental and Safety Laws.
- "ERISA" has the meaning set forth in Section 5.19(a).
- "ERISA Affiliate" has the meaning set forth in Section 5.19(a).
- "Escrow Agent" means Citibank, N.A.
- "Escrow Agreement" has the meaning set forth in Section 3.1(h).
- "Escrow Amount" means \$500,000.
- "Estimated Closing Cash Amount" means an amount equal to \$1,880,000, <u>less</u> the Estimated Closing Indebtedness, <u>less</u> the amount by which the Target Net Working Capital exceeds the Estimated Closing Net Working Capital (if applicable), <u>plus</u> the amount by which the Estimated Closing Net Working Capital exceeds Target Net Working Capital (if applicable).
- "Estimated Closing Indebtedness" means Sellers' good faith estimate and calculation of Closing Indebtedness and each component thereof.

"Estimated Closing Net Working Capital" means Sellers' good faith estimate and calculation of Closing Net Working Capital and each component thereof.

"Estimated Statement" has the meaning set forth in Section 2.2.

"Excluded Liabilities" means any Indebtedness, Liabilities (including but not limited to Environmental Liabilities and Tax Liabilities) or obligations (including but not limited to accounts payable arising prior to the Closing and workers compensation and employers liability obligations with respect to the pre-Closing period) of any nature whatsoever of the Company, its business or of any Seller, whether accrued, absolute, contingent or otherwise, whether known or unknown, whether due or to become due, whether related to the Company's business, whether disclosed on the Disclosure Schedule attached hereto, and regardless of when or by whom asserted: (a) arising out of facts, circumstances or events that occurred or existed prior to the Closing; (b) for infringement or misappropriation of any patent, trademark, copyright, moral right, confidential information, trade secret, or other intellectual property or proprietary right of any other Person occurring prior to or as a result of the Closing; (c) the occurrence or nonoccurrence of any event or circumstance prior to or as a result of the Closing which gives rise to or results in, directly or indirectly, any breach of or default under any Contract at any time; (d) with respect to any product manufactured or sold or service provided prior to the Closing by or on behalf of the Company, any Seller, or any of their predecessors and Affiliates, including any (i) product Liability or similar claim for death, personal injury or damage to or loss of use of any property, (ii) actual or alleged tort, or defect in design, manufacturing, material or workmanship, (iii) actual or alleged breach of any express or implied warranty, (iv) failure to adequately package, label or warn of hazards, and/or (v) mandatory or voluntary recall, retrofit, repair, safety notice or field campaign; and/or (e) relating to the Company's group health insurance plan or any other Company Plan that is, or was at any time prior to the Closing, a multi-employer plan or multiemployer welfare arrangement.

"Fundamental Representation" has the meaning set forth in Section 9.1(a)(ii).

"GAAP" means generally accepted accounting principles in the United States, as in effect from time to time.

"Governing Documents" means the legal document(s) by which any Person (other than an individual) establishes its legal existence or which govern its internal affairs, as amended or restated from time to time. For example, the "Governing Documents" of a limited liability company would be its articles of organization and operating agreement, or similar documents, and "Governing Documents" of a corporation would be its certificate of incorporation and bylaws.

"Government" means any federal, national, foreign, provincial, state or local government, administrative agency or commission, regulatory body, court or other Governmental Authority or instrumentality.

"Governmental Authority" means any foreign or domestic nation, state, province, or other instrumentality or political subdivision thereof (including any county or city), court, administrative agency, commission, regulatory body, and/or any entity exercising executive, legislative, judicial, military, regulatory, administrative or governmental functions thereof.

"<u>Hazardous Materials</u>" means: (a) any material, substance, chemical, waste, product, derivative, compound, mixture, solid, liquid, mineral or gas, in each case, whether naturally occurring or manmade, that is hazardous, acutely hazardous, toxic, or identifiable with words of similar import or regulatory effect under Environmental and Safety Laws; and (b) any petroleum or petroleum-derived products (including, without limitation, gasoline and diesel fuel), radon, radioactive materials or wastes, asbestos containing materials in any form, lead or lead containing materials, urea formal dehyde foam insulation and polychlorinated biphenyls.

"Indebtedness" means, as to the Company, without duplication: (a) any indebtedness for borrowed money of the Company, or any Liability of the Company issued or incurred in substitution, exchange, extension or refinancing of any indebtedness for borrowed money of any Person; (b) any Liability or payment obligation of the Company evidenced by any note, bond, debenture, other debt security, letter of credit or similar instrument; (c) any Liability of the Company for the deferred purchase price of assets, property or services, contingent or otherwise, as obligor or otherwise (including, pursuant to any earn-out, holdback, seller financing, deferred rent and other contingent or deferred consideration); (d) any Contract by which the Company assures a creditor against loss (including in the form of a Contract to repurchase or reimburse); (e) any indebtedness guaranteed in any manner by the Company; (f) any Liabilities under leases that are required to be classified as capitalized lease obligations in accordance with GAAP with respect to which the Company has any Liability, contingently or otherwise, as obligor, guarantor or otherwise; (g) any Liability secured by a Lien on the assets, properties or rights of the Company; (h) any Liability of the Company pursuant to any interest rate, currency, or credit default swap or other hedging arrangement; (i) any deferred revenue of the Company; and (j) all principal, interest, fees, prepayment obligations, penalties, premiums, breakage, termination, make-whole fees or other similar amounts payable with respect to any of the foregoing; provided, that "Indebtedness" shall not include those Closing Current Liabilities incurred by the Company in the Ordinary Course of Business and included in the calculation of Closing Net Working Capital.

"Inventory" means all inventories of the Company owned by the Company and used in the Ordinary Course of Business of the Company, including all finished goods, goods in transit, work in process, samples, packaging materials, raw materials, containers, production and shipping supplies on hand, subject to reserves for damage or obsolescence, and excluding any obsolete, damaged, defective and excessive inventory

<sup>&</sup>quot;Indemnitee" has the meaning set forth in Section 9.4(b).

<sup>&</sup>quot;Indemnitor" has the meaning set forth in Section 9.4(b).

and any other items of inventory that are not useable or saleable in the Ordinary Course of Business of the Company within ninety (90) days following the Closing.

"Knowledge" and terms of similar import mean, with respect to the Company or Sellers, the actual or constructive knowledge of any Seller or any director or officer of the Company or any Subsidiary of the Company, assuming reasonable inquiry as to the matter to which such knowledge relates, and taking into consideration all facts of which such Persons, in the reasonably prudent exercise of their duties, should be aware.

"Latest Balance Sheet" has the meaning set forth in Section 5.4(a).

"Laws" means any statute, law, ordinance, regulation, rule, code, order, constitution, treaty, common law, judgment, decree, other requirement or rule of law of any Governmental Authority.

"Lease" means all leases, subleases, licenses, concessions or other Contracts pursuant to which the Company is granted or obtains any right or interest with respect to real or tangible personal property.

"Leased Real Property" means all land, together with all buildings, structures, improvements, and fixtures located thereon, including all electrical, mechanical, plumbing and other building systems, fire protection, security and surveillance systems, telecommunications, computer, wiring, and cable installations, utility installations, water distribution systems and landscaping, together with all easements and other rights and interests appurtenant thereto (including air, oil, gas, mineral, and water rights) that is subject to any Lease to which the Company is a party, as a tenant or lessee.

"<u>Liability</u>" or "<u>Liabilities</u>" includes, without limitation, liabilities, debts, commitments or other obligations of any nature, whether known or unknown, absolute, accrued, contingent, liquidated, unliquidated, due or to become due, or otherwise.

"Lien" or "Liens" means any lien (statutory or otherwise), hypothecation, encumbrance, claim, Liability, security interest, interest, mortgage, deed of trust, land contract, pledge, restriction, charge, instrument, license, preference, priority, security agreement, easement, covenant, encroachment, option, purchase right, reservation, irregularity, deficiency, default, defect, adverse claim, right of recovery, Tax (including foreign, federal, state and local Tax), order of any Governmental Authority, of any kind or nature, including (a) any conditional sale or other title retention agreement and any lease having substantially the same effect as any of the foregoing or (b) any assignment or deposit arrangement in the nature of a security device, whether secured or unsecured, choate or inchoate, filed or unfiled, scheduled or unscheduled, noticed or unnoticed, recorded or unrecorded, contingent or non-contingent, material or non-material, known or unknown.

"Loss" or "Losses" means any loss, Liability, demand, claim, action, cause of action, cost, damage, diminution in value, deficiency, Tax, penalty, fine or expense, whether or not arising out of third party claims (including interest, penalties, reasonable attorneys' fees and expenses and all amounts paid in investigation, defense or settlement of any of

- "Material Adverse Effect" means any effect or change that is or could be materially adverse to the business, assets, condition (financial or otherwise), operating results, operations or prospects of the Company or to the ability of Sellers to timely consummate the transactions contemplated hereby (regardless of whether or not such adverse effect or change can be or has been cured at any time or whether Buyer has knowledge of such effect or change on the Effective Date).
- "Material Customer" has the meaning set forth in Section 5.23.
- "Material Supplier" has the meaning set forth in Section 5.23.
- "Non-Ordinary Course Liabilities" means Liabilities not incurred in the Ordinary Course of Business of the Company as of the Closing, including: (a) any Liability for severance and other compensation provided or payable to employees or consultants of the Company in connection with the transactions contemplated by this Agreement; (b) any transaction fees and expenses, if any, to be paid by the Company on behalf of any Seller in connection with the transactions contemplated by this Agreement, including financial advisory fees, legal fees and expenses, broker and finder fees and expenses of accountants; (c) any Liability (including increase in obligations) of the Company arising under any Contract or arrangement to which the Company is party arising in whole or in part from the transactions contemplated by this Agreement; (d) any unsatisfied obligation for "withdrawal liability" to a "multiemployer plan" as such terms are defined under ERISA of the Company; (e) any Liability, whether contingent or otherwise, owed by the Company to any Person under any non-competition, consulting or similar arrangements; and (f) any change-of-control or similar payment or increased cost of any Seller or the Company which is triggered in whole or in part by the transactions contemplated by this Agreement.
- "Notice of Disagreement" has the meaning set forth in Section 2.3(b)(iii).
- "Ordinary Course of Business" means the ordinary course of business of the Company, consistent with past practice and the practices of others engaged in a similar line of business as the Company, including with regard to nature, frequency and magnitude.
- "Payoff Letters" has the meaning set forth in Section 3.1(e).
- "Person" means any individual, sole proprietorship, partnership, joint venture, trust, unincorporated association, corporation, limited liability company, entity or Governmental Authority.
- "Pre-Closing Tax Period" has the meaning set forth in Section 8.1.
- "Proprietary Rights" means all of the following which are owned by, issued to, held, licensed or used by the Company or otherwise used in the business of the Company: (a) U.S. and foreign patents, patent applications, patent disclosures, patent rights and inventions (whether or not patentable and whether or not reduced to practice); (b) trademarks, trade names, service marks and trade dress (including any and all rights in the Company Marks), tradether with all acceptable associated therewith all translations.

adaptations, derivations and combinations of the foregoing, and all logos related to the foregoing; (c) copyrights and copyrighted works; (d) internet domain names; and (e) all registrations and applications related to any of the foregoing; (f) the right to obtain renewals, extensions, substitutions, continuations, continuations-in-part, divisions, reissues, re-examinations or similar legal protections for any of the foregoing; (g) trade secrets and other Confidential Information, including ideas, know-how, related processes and techniques, research and development information, drawings, specifications, designs, plans, proposals and technical data and manuals; (h) computer software and website content (including data, object code, source code and related documentation, whether stored locally or hosted by a third party); and (i) all other intangible properties and rights relating to the Company or its business.

- "Pro Rata Share" means (i) with respect to Dustin Baker, 80%; and (ii) with respect to Brant Baker, 20%.
- "Purchase Price" means an aggregate amount equal to: (i) Closing Consideration, as adjusted by the Adjustment Amount, plus (ii) the amount of the Liabilities of the Company.
- "Remediation" means any investigation, monitoring, maintenance of engineering or institutional controls, removal, corrective action, response action, mitigation, treatment, decontamination or cleanup of Hazardous Materials, waste material, radioactive material, pollution or contamination present or alleged to be present in the Environment, soil, sediment, groundwater, surface water or other environmental medium or the indoor environment.
- "Securities Laws" means the U.S. Securities Act of 1933, the U.S. Securities and Exchange Act of 1934, or any other federal, state or local laws relating to the offer, issuance, transfer, sale, resale or other distribution of securities or regulations promulgated thereunder.
- "Seller Indemnified Parties" means each Seller and their respective successors and assigns.
- "Seller Representative" has the meaning set forth in Section 10.12(a).
- "Straddle Period" has the meaning set forth in Section 8.2.
- "Subsidiary" means, with respect to any Person, any corporation, partnership, limited liability company, association or other business entity of which: (a) if a corporation, a majority of the total voting power of shares of stock entitled (irrespective of whether, at the time, stock of any other class or classes of such corporation shall have or might have voting power by reason of the happening of any contingency) to vote in the election of directors, managers or trustees thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more of the other Subsidiaries of that Person or a combination thereof; or (b) if a partnership, limited liability company, association or other business entity, either (i) a majority of the partnership, membership or other similar

that Person or one or more Subsidiaries of that Person or a combination thereof, or (ii) such Person is a general partner, manager or director of such partnership, limited liability company, association or other entity.

"Target Net Working Capital" means \$600,000.00.

"Tax" or "Taxes" means any federal, state, local or foreign income, gross receipts, license, payroll, employment, excise, severance, stamp, occupation, premium, property (including general and special real estate taxes and assessments, special service area charges, tax increment financing, charges, payments in lieu of taxes and similar charges and assessments), windfall profits, environmental (including tax under Code Section 59A), customs duties, capital stock, franchise, margin, profits, withholding, foreign or domestic withholding, social security (or similar), unemployment, disability, real property, personal property, sales, use, transfer, registration, value added, alternative or add-on minimum, estimated, escheat or unclaimed property, or other tax, governmental fee, governmental assessment or governmental charge of any kind whatsoever, whether computed on a separate or consolidated, unitary or combined basis or in any other manner including any interest, penalties or additions to Tax or additional amounts with respect to the foregoing whether disputed or not.

"Tax Returns" means returns, declarations, reports, claims for refund, information returns or other documents (including any related or supporting schedules, statements or information) filed or required to be filed in connection with the determination, assessment or collection of Taxes of any party or the administration of any Laws relating to any Taxes.

"Third Party Claim" has the meaning set forth in Section 9.4(b).

"Transfer Taxes" means any sales, use, goods and services, harmonized sales, stock transfer, real property transfer, transfer, stamp, registration, documentary, recording or similar Tax incurred in connection with the transactions contemplated hereby, together with any interest, penalties or additions to such Tax or additional amounts with respect thereto, whether disputed or not.

"Willful Breach" means any willful or intentional breach or violation of any of the representations, warranties, covenants or obligations set forth in this Agreement.

## **ARTICLE II**

## PURCHASE AND SALE OF MEMBERSHIP INTERESTS

2.1 Purchase and Sale of Membership Interests. Upon the terms and subject to the satisfaction or waiver of the conditions of this Agreement, at the Closing, Buyer shall purchase, acquire and accept from each Seller, and each Seller shall sell, transfer, assign and deliver to Buyer, all of the Membership Interests owned by such Seller, free and clear of all Liens, which Membership Interests will collectively constitute as of the Closing all of the outstanding membership interests or other limited liability company interests of the

- 2.2 <u>Estimates</u>. At least three (3) Business Days prior to the Effective Date, Sellers shall deliver to Buyer a written statement (the "<u>Estimated Statement</u>") setting forth Sellers' good faith estimation and calculation of each of the following amounts as of the Closing: (a) the Estimated Closing Net Working Capital and each component thereof; (b) the Estimated Closing Indebtedness; and (c) the Estimated Closing Cash Amount. Sellers shall provide Buyer supporting documentation and explanation for the estimates, assumptions and calculations reflected on the Estimated Statement upon request.
- Closing Consideration; Adjustment Amount; Escrow.
  - (a) <u>Estimated Closing Consideration</u>. In exchange for all of the Membership Interests, at Closing, Buyer shall deliver the Closing Consideration as follows:
    - (i) Buyer shall deliver each Seller such Seller's Pro Rata Share of the Estimated Closing Cash Amount, by wire transfer of immediately available funds to a U.S. bank account designated by such Seller in writing not less than three (3) Business Days prior to the Effective Date. The Estimated Closing Cash Amount is subject to the adjustments contemplated by Section 2.3(b).
    - (ii) Buyer shall deliver to each Seller such Seller's Pro Rata Share of the Closing Stock Consideration, bearing such restrictive legends as may be required by applicable Law, The NASDAQ Stock Market rules and this Agreement. In the event the calculation of the total number of shares of Buyer Stock included in the Closing Stock Consideration or the calculation of any Seller's Pro Rata Share thereof results in any fractional shares of Buyer Stock, Buyer shall have the option to deliver cash in lieu of such fractional shares of Buyer Stock, in an amount equal to the Buyer Stock Closing Price, multiplied by such fractional share of Buyer Stock, by wire transfer of immediately available funds to the U.S. bank account designated by such Seller pursuant to Section 2.3(a)(i).
    - (iii) Buyer shall deliver the Escrow Amount to the Escrow Agent, by wire transfer of immediately available funds to a U.S. bank account designated by the Escrow Agent not less than three (3) Business Days prior to the Effective Date.
  - (b) Working Capital and Other Adjustment to the Closing Cash Amount.
    - (i) Following the Closing, in accordance with this <u>Section 2.3(b)</u>, the Closing Cash Amount shall be: (A) reduced by the amount, if any, by which the Estimated Closing Net Working Capital exceeds the Closing Net Working Capital; or (B) increased by the amount, if any, by which the Closing Net Working Capital exceeds the Estimated Closing Net Working Capital.
    - (ii) Without duplication of any adjustment made pursuant to <u>Section 2.3(b)(i)</u>, the Closing Cash Amount shall be further: (A) decreased, dollar for dollar, by the amount of any Closing Indebtedness not included in the calculation of the Estimated Closing Indebtedness and the Estimated Closing Cash Amount; and (B) without duplication of any decrease under clause (A),

Non-Ordinary Course Liabilities. The net amount of all sums that increase or decrease the Closing Cash Amount as contemplated by <u>Section 2.3(b)(i)</u> and this <u>Section 2.3(b)(ii)</u>, as finally determined pursuant to this <u>Section 2.3(b)</u>, is referred to herein as the "<u>Adjustment Amount</u>."

- (iii) Within ninety (90) days following the Closing, Buyer will prepare and deliver to the Seller Representative a statement (the "Adjustment Statement") setting forth its good faith calculation of: (A) the Closing Net Working Capital; (B) the Excluded Liabilities and the Non-Ordinary Course Liabilities, if any; (C) the Closing Indebtedness, if any; and (D) the Adjustment Amount. The Adjustment Statement and the calculation of the Adjustment Amount set forth therein shall become final and binding upon the Parties on the thirtieth (30th) day following delivery thereof by Buyer, unless the Seller Representative gives written notice of its disagreement (a "Notice of Disagreement") to Buyer prior to such date. During such period, the Seller Representative shall have reasonable access, upon reasonable prior notice and during normal business hours, to those books and records of the Company relevant to the calculation of the Adjustment Amount, solely for purposes reasonably related to Seller's verification of the amounts and calculations set forth in the Adjustment Statement. Any Notice of Disagreement shall specify in reasonable detail: (V) the nature and amount of any disagreement so asserted; (W) Sellers' calculation of the Closing Net Working Capital; (X) Sellers' calculation of the Excluded Liabilities and the Non-Ordinary Course Liabilities; (Y) Seller's calculation of the Closing Indebtedness; and (Z) Sellers' calculation of the Adjustment Amount.
- (iv) If a timely Notice of Disagreement is received by Buyer, then the Adjustment Statement and the calculation of the Adjustment Amount set forth therein shall become final and binding upon the Parties on the earlier of: (A) the date Buyer and the Seller Representative resolve in writing any differences they have with respect to any matter specified in the Notice of Disagreement; or (B) the date any matters properly in dispute are finally resolved in writing by the Accounting Firm pursuant to Section 2.3(b)(v). During the thirty (30) days immediately following the delivery of a Notice of Disagreement, Buyer and the Seller Representative shall consult in good faith to resolve in writing any differences they may have with respect to any matter specified in the Notice of Disagreement. During such period, Buyer shall have full access to the work papers of Sellers and the Seller Representative prepared in connection with the Notice of Disagreement.
- (v) At the end of such thirty (30) day negotiation period described in <u>Section 2.3(b)(iv)</u>, at the request of any Party, any and all matters which remain in dispute and which were included in the Notice of Disagreement shall be submitted to the Salt Lake City, Utah office of BDO USA, LLP (the "Accounting Firm") for binding resolution. The fees and expenses of the

Accounting Firm shall be borne by Buyer, on the one hand, and Sellers, on the other hand, in the proportion that the aggregate dollar amount of items submitted to the Accounting Firm that are unsuccessfully disputed by Buyer and the Seller Representative, as applicable, bears to the aggregate dollar amount of all items submitted to the Accounting Firm. The Accounting Firm shall determine and report in writing to Buyer and the Seller Representative as to the resolution of all disputed matters submitted to the Accounting Firm and the effect of such determinations on the Adjustment Statement within ten (10) Business Days after such submission or such longer period as the Accounting Firm may reasonably require, and such determinations shall be final, conclusive and binding as to the Parties. In resolving any disputed item, the Accounting Firm, acting in the capacity of an expert and not as an arbitrator: (A) shall limit its review to matters specifically set forth in the Notice of Disagreement as a disputed item; and (B) shall not assign a value to any item greater than the greatest value for such item claimed by any Party or less than the smallest value for such item claimed by any other Party in the Adjustment Statement or in the Notice of Disagreement. Buyer and the Seller Representative shall each furnish to the Accounting Firm such work papers and other documents and information relating to the disputed issues, and will provide interviews and answer questions, as the Accounting Firm may reasonably request in connection with its determination of such disputed items.

- (vi) If the Adjustment Amount is an increase to the Closing Cash Amount, Buyer will make (or cause to be made) a payment to each Seller an amount equal to such Seller's Pro Rata Share of the Adjustment Amount, by wire transfer or delivery of other immediately available funds to the U.S. bank account designated by such Seller pursuant to Section 2.3(a)(i), within five (5) Business Days of the final determination of the Adjustment Amount. Notwithstanding the foregoing, Buyer shall be entitled to withhold and set off any amount due to Sellers pursuant to this Section 2.3(b)(vi) against any amount Buyer reasonably believes is due to any of the Buyer Indemnified Parties from any Seller pursuant to this Agreement.
- (vii) If the Adjustment Amount is a decrease to the Closing Cash Amount, Sellers will make (or cause to be made) payment of the Adjustment Amount to Buyer by wire transfer or delivery of other immediately available funds, to an account designated by Buyer to the Seller Representative, within five (5) Business Days of the final determination of the Adjustment Amount. Buyer shall also have the right (but not the obligation) to withhold any other amount due from Buyer to Sellers and set off such amount against any amount due from Sellers to Buyer pursuant to this Section 2.3(b)(vii).
- (viii) Any payments required under this <u>Section 2.3(b)</u> shall accrue interest at the Applicable Rate per annum from the Effective Date until paid.

- (c) <u>Escrow</u>. The Escrow Agent will hold the Escrow Amount pursuant to the Escrow Agreement to secure and be available on a non-exclusive basis for the payment of the Liabilities of any Seller arising out of or relating to this Agreement (including, but not limited to, any Losses to which any Buyer Indemnified Party may be entitled pursuant to Section 9.2(a)).
- 2.4 Closing. The closing of the transactions contemplated by this Agreement (the "Closing") shall take place on the Effective Date at the offices of Buyer at 312 South 1250 West, Lindon, Utah 84042 or by exchange of documents by electronic transmission, by a notification method permitted pursuant to Section 10.2; provided, that the Closing shall be deemed effective as of the opening of business on the Effective Date.

# ARTICLE III CLOSING DELIVERABLES

- 3.1 <u>Closing Deliverables of Sellers</u>. At the Closing, Sellers shall deliver (or cause to be delivered) to Buyer:
  - (a) assignments of the Membership Interests held by each Seller, duly endorsed by such Seller in blank and sufficient to transfer the Membership Interests and any certificates representing such Membership Interests to Buyer free and clear of all Liens, together with any original certificates representing such Membership Interests;
  - (b) all legally binding consents, approvals, authorizations and waivers from any Governmental Authority or any other Person that are: (i) necessary for consummation of the transactions contemplated hereby by the Company and any Seller; or (ii) required pursuant to any Contract of the Company, including, without limitation, (A) to avoid and breach of or default under such Contract as a result of the transactions contemplated by this Agreement (with or without the giving of notice, the passage of time or both), or (B) necessary for such Contract to remain in full force and effect following the Closing, without any modification to the terms or conditions of such Contract (with or without the giving of notice, the passage of time or both), or (C) in connection with any assignment of any such Contract to, or merger of the Company with or into, Buyer or is Affiliates as of or promptly following the Closing;
  - (c) a certificate, duly executed by Sellers as members of the Company, certifying as to: (i) the most recent Articles of Organization of the Company and any amendments thereto (certified as of a recent date by the Ohio Secretary of State); (ii) a good standing certificate for the Company, issued on, or not more than three (3) days before, the Effective Date by the Ohio Secretary of State; and (iii) the most recent operating agreement or limited liability company agreement of the Company and any amendments thereto;

- (d) one or more certificates of each Seller and the Company complying with Treasury Regulations Section 1.1445-2, establishing that the purchase of the Membership Interest is exempt from withholding under Section 1445 of the Code;
- (e) with respect to each Lien affecting the Membership Interests or any of the assets or properties of the Company, and each item of Closing Indebtedness or Non-Ordinary Course Liabilities, of whatever nature, either: (i) evidence satisfactory to Buyer and its counsel of the payoff, satisfaction, discharge and the effective release of such Lien, item of Closing Indebtedness and/or Non-Ordinary Course Liabilities; or (ii) an effective payoff and release letter with respect to such Lien, item of Closing Indebtedness and/or Non-Ordinary Course Liabilities that are not released and discharged prior to the Closing (collectively, the "Payoff Letters"), which Payoff Letters shall each (A) be in a form satisfactory to Buyer and its counsel, (B) be executed by the holders of such Lien, item of Closing Indebtedness and/or Non-Ordinary Course Liabilities in favor of the Company and Buyer, (C) set forth the outstanding balance of such Lien, item of Closing Indebtedness and/or Non-Ordinary Course Liabilities subject to such Payoff Letter as of the Effective Date that is not subject to adjustment or recalculation if paid on the Effective Date, (D) provide payment and wire transfer instructions applicable thereto, and (E) provide for the full and final payment, satisfaction, release and discharge of such Lien, item of Closing Indebtedness and/or Non-Ordinary Course Liabilities subject to such Payoff Letter and authorize the Company and Buyer to file terminations (including UCC termination statements) with respect to such Liens upon delivery of the payments required pursuant to the applicable Payoff Letter;
- (f) written resignations of each manager and officer of the Company, effective as of or prior to the Closing;
- (g) the original books of account, minute books, minutes and other records of the Company, the official seal, if any, of the Company and such other documents, records, keys and other items as shall be reasonably necessary for the operation of the business of the Company;
- (h) counterparts to an Escrow Agreement in the form attached hereto as <u>Exhibit</u>
   3.1(h) (the "<u>Escrow Agreement</u>"), duly executed by the Seller Representative and the Escrow Agent;
- (i) an at-will employment offer letter in the form attached hereto as <a href="Exhibit 3.1(i)">Exhibit 3.1(i)</a> (the "Dustin Offer Letter"), duly executed by Dustin Baker; and
- a counterpart to an at-will employment offer letter in the form attached hereto as <u>Exhibit 3.1(j)</u> (the "<u>Brant Offer Letter</u>"), duly executed by Brant Baker.

All actions to be taken and deliveries to be made by Sellers in connection with the consummation of the transactions contemplated hereby, and all certificates, instruments and other documents required to effect the transactions contemplated hereby, each as

reasonably requested by Buyer, shall be reasonably satisfactory in form and substance to Buyer. The timely delivery of each of the items identified in this <u>Section 3.1</u> is a condition precedent to the obligation of Buyer to proceed with the Closing, and may be waived only if, and to the extent, expressly set forth in writing signed by Buyer specifying in reasonable detail the provision being waived.

- 3.2 <u>Closing Deliveries of Buyer</u>. At the Closing, Buyer shall deliver (or cause to be delivered):
  - (a) the Closing Consideration, pursuant to Section 2.3(a); and
  - (b) a counterpart to the Escrow Agreement, duly executed by Buyer;
  - a counterpart to the Dustin Offer Letter, duly executed by the Company to become effective immediately following the Closing; and
  - (d) a counterpart to the Brant Offer Letter, duly executed by the Company to become effective immediately following the Closing.

All actions to be taken and deliveries to be made by Buyer in connection with the consummation of the transactions contemplated hereby, and all documents required to be delivered by Buyer to effect the transactions contemplated hereby, each as reasonably requested by the Seller Representative, shall be reasonably satisfactory in form and substance to the Seller Representative. The timely delivery of each of the items identified in this Section 3.2 is a condition precedent to the obligation of Sellers to proceed with the Closing, and may be waived only if, and to the extent, expressly set forth in writing signed by the Seller Representative specifying in reasonable detail the provision being waived.

#### ARTICLEIV

## REPRESENTATIONS AND WARRANTIES OF SELLERS

As an inducement to Buyer to enter into this Agreement and consummate the transactions contemplated hereby, each Seller hereby severally, but not jointly, represents and warrants to Buyer that all of the following representations and warranties in this <u>ARTICLE IV</u> are true and correct as of the Effective Date and as of the Closing with respect to such Seller:

- 4.1 <u>Power of Sellers.</u> Such Seller has all requisite power, authority and capacity to execute and deliver this Agreement and to consummate the transactions contemplated hereby.
- Authorization; No Breach. The execution, delivery and performance of this Agreement and the other agreements to be executed and delivered by such Seller pursuant to this Agreement, and the consummation of the transactions contemplated hereby and thereby, have been duly and validly authorized by all requisite action on the part of such Seller, and no other action on the part of such Seller are necessary to authorize the execution, delivery or performance of this Agreement or the other agreements contemplated hereby.

  This Agreement, and the other agreements contemplated hereby to be executed and

delivered by such Seller, constitute valid and binding obligations of such Seller, enforceable against such Seller in accordance with their respective terms, except to the extent such enforceability may be limited by bankruptcy, insolvency or similar laws, or by general principles of equity, applicable to creditors' rights generally. The execution, delivery and performance of this Agreement and the other agreements contemplated hereby to be executed and delivered by such Seller, and the consummation of the transactions contemplated hereby and thereby, do not and shall not: (a) conflict with, result in any breach or violation of, constitute a default under, give any third party the right to terminate or to accelerate any obligation under, or require any authorization, consent, approval, exemption, notice or other action under, the provisions of any Contract to which such Seller is a party, or by which such Seller or the Membership Interests of such Seller are bound or affected, or any Law to which such Seller or the Membership Interests of such Seller are subject; (b) result in the creation of any Lien of any kind upon or with respect to any of the Membership Interests of such Seller; or (c) require any authorization, consent, approval, exemption from or other action by, or notice to or filing with, any Governmental Authority.

- 4.3 Brokerage. There are no claims against such Seller, and such Seller has no Liability, for brokerage commissions, finders' fees or similar compensation in connection with the transactions contemplated by this Agreement based on any Contract or arrangement made by or on behalf of such Seller.
- A.4 Title to Membership Interests. Such Seller is the holder of record and beneficially owns all of the Membership Interests set forth with respect to such Seller in Section 5.3 of the Disclosure Schedule, and collectively Sellers are the holders of record and beneficially own all of the Membership Interests. Such Seller has good, valid and marketable title to all of the Membership Interests of such Seller and such Membership Interests shall be, as of the Closing, free and clear of all Liens. Such Seller is not a party to, and the Membership Interests of such Seller are not subject to, any: (a) convertible securities, calls, preemptive rights, options, warrants, purchase rights, rights of refusal, or other Contracts or commitments (other than this Agreement) that would require such Seller or any other Person to sell, transfer or otherwise dispose of the Membership Interests of such Seller, or prohibit or conflict with such Seller's obligations under this Agreement; or (b) proxies, voting rights, shareholder agreements or other Contracts or understandings with respect to the voting or transfer of the Membership Interests of such Seller.

## 4.5 Investment.

(a) Such Seller has been advised that the shares of Buyer Stock included in the Closing Stock Consideration have not been registered under Securities Laws. Such Seller acknowledges that the shares of Buyer Stock included in the Closing Stock Consideration are being issued pursuant to a specific exemption from the registration requirements of applicable Securities Laws, and that Buyer's reliance on such exemption depends, in part, on such Seller's representations and warranties set forth in this Section 4.5.

- (b) Such Seller acknowledges and understands that no portion of the shares of Buyer Stock included in the Closing Stock Consideration can be offered, sold, resold, transferred or distributed: (i) unless or until such shares of Buyer Stock included in the Closing Stock Consideration are registered under applicable Securities Laws, or exemptions from such registration requirements are available and complied with as to such offer, sale, resale, transfer or distribution; (ii) until the expiration of the twelve (12)-month lock up provided in Section 7.7; and (iii) except in compliance with all Securities Laws.
- (c) Such Seller acknowledges and understands that: (i) the receipt of the shares of Buyer Stock included in the Closing Stock Consideration is a long-term investment; (ii) such Seller must bear the economic risk of investment in such shares of Buyer Stock for an indefinite period of time; and (iii) such Seller may not be able to liquidate its investment in such shares of Buyer Stock in the event of an emergency or pledge any of such shares of Buyer Stock as collateral for loans or other obligations.
- (d) Such Seller is entering into this Agreement and will receive the shares of Buyer Stock included in the Closing Stock Consideration for such Seller's own account for investment, not as a nominee or agent, and not with a view to, or for resale in connection with, any distribution thereof. Such Seller does not have any contract, undertaking, agreement or arrangement with any Person to sell, transfer or grant participation to such Person or to any third Person in or to, any of such shares of Buyer Stock.
- (e) Such Seller is familiar with and understands the business, operations and prospects of Buyer. Such Seller acknowledges that Buyer has: (i) provided such Seller access to the public filings of Buyer and to all information in Buyer's possession relating to the shares of Buyer Stock included in the Closing Stock Consideration, Buyer and its business; (ii) made the directors, officers and representatives of Buyer available for interview by such Seller; (iii) provided to such Seller the opportunity to ask questions and receive answers concerning the terms and conditions of the issuance of the shares of Buyer Stock included in the Closing Stock Consideration as set forth in this Agreement; and (iv) furnished such Seller with all documents and other information required for or otherwise requested by such Seller to make an informed decision with respect to entering into this Agreement and investment in such shares of Buyer Stock.
- (f) Such Seller has such knowledge and experience in financial and business matters that such Seller is capable of evaluating the merits and risks of such Seller's investment in the shares of Buyer Stock included in the Closing Stock Consideration. Such Seller has obtained, to the extent such Seller deems necessary, his own professional advice with respect to the risks inherent in the investment in the such shares of Buyer Stock, and the suitability of an investment in such shares of Buyer Stock in light of such Seller's financial condition and investment needs. An investment in such shares of Buyer Stock is suitable for such Seller based upon such Seller's investment objectives and financial needs.

- (g) Such Seller acknowledges that such Seller's investment in the shares of Buyer Stock included in the Closing Stock Consideration is a speculative investment that has limited liquidity, involves a high degree of risk and is subject to the risk of complete loss. Such Seller is in a financial position to hold such shares of Buyer Stock for an indefinite period without impairing such Seller's financial condition and is able to bear the economic risk and withstand a complete loss of such Seller's investment in such shares of Buyer Stock. Such Seller has adequate means for providing for such Seller's current financial needs and contingencies and has no need for liquidity of investment with respect to such shares of Buyer Stock.
- (h) Such Seller is a resident of the state set forth with respect to such Seller on Section 5.3 of the Disclosure Schedule.
- 4.6 <u>Disclosure</u>. No representation or warranty set forth in this <u>ARTICLE IV</u> contains any untrue statement of a material fact or omits a material fact necessary to make the statements contained herein, in light of the circumstances in which they were made, not misleading.

#### **ARTICLE V**

## REPRESENTATIONS AND WARRANTIES REGARDING THE COMPANY

As an inducement to Buyer to enter into this Agreement and consummate the transactions contemplated hereby, Sellers hereby represent and warrant to Buyer that all of the following representations and warranties in this <u>ARTICLE V</u> are true and correct as of the Effective Date and as of the Closing:

5.1 Organization and Power. The Company is a limited liability company duly organized, validly existing and in good standing under the laws of the State of Ohio. The Company has obtained and currently maintains all qualifications to do business as a foreign limited liability company in all other jurisdictions in which the character of the Company's properties or the nature of the Company's activities require it to be so qualified, except where the failure to obtain or maintain such qualification would not have a Material Adverse Effect. All jurisdictions where the Company is qualified to do business are set forth on Section 5.1 of the Disclosure Schedule. The Company has all requisite power and authority and all authorizations, licenses and permits necessary to own and operate its assets and to conduct its business as now conducted and as proposed to be conducted. The copies of the Company's Governing Documents, which have been furnished to Buyer, are correct and complete. The books of account, member ledgers, minute books and other records of the Company are accurate, up to date and complete, and have been maintained in accordance with prudent business practices and all applicable Laws. The Company is not in default under or in breach or violation of any provision of its Governing Documents or any currently effective resolution or other action adopted by the Company's board of directors or shareholders. Section 5.1 of the Disclosure Schedule lists the current directors and officers of the Company.

5.2 Authorization; No Breach. No actions on the part of the Company, whether by the shareholders, directors or otherwise, are necessary to approve and authorize the execution, delivery and performance of this Agreement and the other agreements contemplated hereby, or the consummation of the transactions contemplated hereby and thereby. Except as set forth on Section 5.2 of the Disclosure Schedule, neither the execution and delivery of this Agreement nor any of the other agreements contemplated hereby, nor the consummation of the transactions contemplated hereby or thereby, will, directly or indirectly (with or without notice, lapse of time or both): (a) conflict with or result in any breach of, constitute a default under, give any third party the right to terminate or to accelerate any Liability or other obligation under, result in the amendment or modification of any term or condition of, or require any authorization, consent, approval, exemption, notice or other action under the provisions of the Governing Documents of the Company, or under the provisions of any Contract by which the Company or any of its assets are bound or affected, or under any license, permit, certificate, accreditation or other authorization of the Company, or under any Law to which the Company is subject; (b) result in the creation of any Lien of any kind upon any of the Membership Interests or any of the Company's properties, assets or rights; or (c) require any authorization, consent, approval, exemption or other action by, or notice to or filing with, any Governmental Authority. Other than as set forth on Section 5.2 of the Disclosure Schedule, no notice to, filing with, or consent of, any Person is necessary in connection with, and no "change of control" provision in any Contract is triggered by, the execution, delivery or performance of this Agreement and the other agreements contemplated hereby, nor the consummation of the transactions contemplated hereby or thereby. All notices, filings and consents listed on Section 5.2 of the Disclosure Schedule have been given, made and obtained, and are in full force and effect.

# 5.3 Capitalization; Subsidiaries.

All of the Membership Interests are owned, of record and beneficially, by Sellers (a) as set forth on Section 5.3 of the Disclosure Schedule. The Membership Interests are denominated in uncertificated units, and are the only membership, limited liability company, capital, economic or other interests that have been issued by the Company and, upon transfer of the Membership Interests to Buyer pursuant to this Agreement, Buyer shall be admitted as the sole member of the Company and shall hold all of the Membership Interests. All of the Membership Interests have been duly authorized and validly issued, and are fully paid and non-assessable. There are no: (i) outstanding obligations, options, warrants, convertible or exchangeable securities or other rights, agreements or commitments (contingent or otherwise) relating to any interest in the Company (including any membership, limited liability company, equity, capital, economic or other interests) or that could require the Company to issue or sell or otherwise cause to become outstanding any such interests; (ii) outstanding obligations of the Company to repurchase, redeem or otherwise acquire any Membership Interests or other interest in the Company (including any membership, limited liability company, equity, capital, economic or other interests) in the Company or to make any investment (in the form of a loan, capital contribution or otherwise) in any other Person: (iii) voting truste charaholder agreements preemptive rights of

first refusal, restrictions on transfer, proxies or other agreements or understandings in effect with respect to the Membership Interests or other interest in the Company (including any membership, limited liability company, equity, capital, economic or other interests) or the voting or transfer thereof; (iv) outstanding or authorized appreciation rights, phantom equity, profit participation or similar rights with respect to the Company, the Membership Interests or other interest in the Company (including any membership, limited liability company, equity, capital, economic or other interests); or (v) certificates or other negotiable instruments that were issued or granted by the Company to represent or evidence the Membership Interests.

- (b) The Company does not have any Subsidiaries. The Company does not own or hold the right to acquire any shares of stock or any other security or interest in any other Person, and does not have any obligation to make any investment in any Person.
- 5.4 <u>Financial Statements</u>. <u>Section 5.4</u> of the Disclosure Schedule sets forth the following financial statements of the Company (the "<u>Financial Statements</u>"):
  - the unaudited balance sheet of the Company as of June 30, 2019 (the "<u>Latest Balance Sheet</u>"), and the related statements of income, members' equity and cash flows (or the equivalents) for the interim period then ended; and
  - (b) the unaudited balance sheet of the Company as of December 31, 2016, 2017 and 2018, and the related statements of income, members' equity and cash flows (or the equivalents) for the years then ended.

Each of the Financial Statements (including in all cases the notes thereto) is accurate and complete, is consistent with the books and records of the Company (which, in turn, are accurate and complete), presents fairly the financial condition, material transactions and results of operations of the Company as of the respective dates thereof and the operating results of the Company for the periods covered thereby, and have been prepared in accordance with the accounting methods and practices of the Company, consistently applied, in the Ordinary Course of Business and in a manner consistent with the preparation of the Company's Tax Returns. The Company does not have any material off-balance sheet Liabilities or financing arrangements.

Absence of Undisclosed Liabilities. Except as set forth on Section 5.5 of the Disclosure Schedule, the Company does not have, nor will it have as of the Closing, any Liabilities (whether accrued, absolute, contingent, un-liquidated or otherwise, whether due or to become due, and regardless of when or by whom asserted) at or as of the Closing or arising out of transactions entered into at or prior to the Closing, or arising out of any action or inaction at or prior to the Closing, or to the Knowledge of Sellers, any facts or circumstances existing at or prior to the Closing, including Taxes with respect to or based upon periods, transactions or events occurring on or before the Closing, except: (a) obligations under Contracts described on Section 5.13(a) of the Disclosure Schedule (but not Liabilities for breach or violation of any such Contract occurring on or prior to the

- Closing); (b) Liabilities reflected on the Latest Balance Sheet; and (c) Liabilities which have arisen after the date of the Latest Balance Sheet in the Ordinary Course of Business (none of which is a Liability for breach of any Contract, breach of warranty, tort, product liability, infringement, claim or lawsuit).
- 5.6 Accounts Receivable. Section 5.6 of the Disclosure Schedule provides an accurate and complete breakdown and aging of all accounts receivable, notes receivable and other receivables of the Company as of the Closing. Except as set forth in Section 5.6 of the Disclosure Schedule, all accounts and notes receivable of the Company reflected on the Latest Balance Sheet and all accounts and notes receivable to be reflected on the Adjustment Statement and to be taken into account in the calculation of Closing Net Working Capital (net of allowances for doubtful accounts as reflected thereon and as determined in accordance with GAAP) are valid, currently due and payable receivables arising in the Ordinary Course of Business and are and shall be current and collectible at the aggregate recorded amount therefor as shown on the Latest Balance Sheet and on the Adjustment Statement, as the case may be (net of allowances for doubtful accounts as reflected thereon and as determined in accordance with GAAP). Since the Balance Sheet Date, except as set forth in Section 5.6 of the Disclosure Schedule, the Company has not changed the manner or timing for issuance of invoices or the collection of debts or receivables. No Person has any Liens on such receivables or any part thereof, and no agreement for deduction, credit, free goods, discount or other deferred price or quantity adjustment has been made, offered or promised with respect to any such receivables.
- 5.7 Accounts Payable. Section 5.7 of the Disclosure Schedule sets forth a list of all accounts payable of the Company as of the date of this Agreement, together with the name of each payee, the date each such payment is due, and the nature of the transaction in which such payable was incurred (if other than a trade account payable incurred in the Ordinary Course of Business). Since the Balance Sheet Date, Company has paid its creditors and suppliers substantially within the times agreed with such creditors, except where charges are being disputed in good faith.
- No Material Adverse Change. From the Balance Sheet Date through the date of this Agreement, to the Knowledge of Sellers there has not occurred or existed any change, event, fact, circumstance or effect that has had a Material Adverse Effect or that would, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. Since the Balance Sheet Date, the Company has conducted its business only in the Ordinary Course of Business.
- 5.9 <u>Absence of Certain Developments</u>. Except as set forth on <u>Section 5.9</u> of the Disclosure Schedule, since the Balance Sheet Date, the Company has not:
  - (a) (i) issued, sold, transferred, pledged, disposed of or suffered any Lien on any of its the Membership Interests or other interest in the Company (including any membership, limited liability company, equity, capital, economic or other interests) or any securities or rights convertible, exchangeable or exercisable into such interests; (ii) granted any options, warrants or other rights to purchase or obtain any such interests; (iii) split, combined, subdivided, reclassified or

- recapitalized any of its such interests; (iv) declared, set aside or paid any dividend or distribution; or (v) redeemed, purchased or otherwise acquired any such interest:
- (b) amended, modified, restated or amended and restated any of its Governing Documents;
- adopted a plan or agreement of liquidation, dissolution, restructuring, merger, consolidation, recapitalization or other reorganization;
- (d) made any distributions or other payments to Sellers or to any Affiliate of the Company or Sellers, except the payment of guaranteed payments and benefits with respect to services rendered by Sellers to the Company in the Ordinary Course of Business;
- (e) (i) paid any accounts payable other than in the Ordinary Course of Business, or (ii) delayed or postponed the payment of any accounts payable or commissions or any other Liability, or (iii) agreed or negotiated with any party to extend the payment date of any trade or accounts payable or commission or any other Liability, or (iv) accelerated the collection of (or discounted) any accounts or notes receivable (whether billed or unbilled) or accelerated any deferred revenue, or (v) otherwise taken any actions or omitted to take any actions with the intent or the purpose of increasing Closing Net Working Capital or satisfying the Target Net Working Capital as of the Closing;
- (f) delayed issuing any checks or payments;
- (g) paid or refinanced any Liability, other than in the Ordinary Course of Business;
- (h) mortgaged, pledged or subjected to any Lien, any of its properties, assets or rights;
- sold, leased, assigned or transferred any of its tangible assets, except in the Ordinary Course of Business, or canceled without fair consideration any debts or claims owing to or held by it;
- (i) sold, assigned, licensed, sublicensed, transferred or encumbered any
  Proprietary Rights or other intangible assets, or (ii) disclosed any proprietary
  Confidential Information to any Person (other than Buyer and Buyer's
  representatives, agents, attorneys and accountants), or (iii) abandoned or
  permitted to lapse any Proprietary Rights;
- (k) (i) made or granted any bonus, or any wage, salary or compensation increase, to any employee or independent contractor, or any group of employees or independent contractors, except as required by pre-existing contracts or, in the case of non-officer employees, consistent with past practice, or (ii) made or granted any increase in any employee benefit plan or arrangement, or (iii) amended or terminated any existing employee benefit plan or arrangement or

- adopted any new employee benefit plan or arrangement, or (iv) entered into, modified or terminated any collective bargaining agreement or relationship;
- (I) incurred any Indebtedness or incurred or become subject to any material Liability, except current Liabilities incurred in the Ordinary Course of Business and Liabilities under Contracts entered into in the Ordinary Course of Business (none of which is a Liability for breach of any Contract, breach of warranty, tort, product liability, infringement, claim or lawsuit);
- (m) made any loans or advances to, or guarantees for the benefit of, any Person;
- (n) made any write down in the value of its inventory or work in process, or made any write off of any accounts receivables;
- suffered any extraordinary Losses or waived any material rights or claims, whether or not in the Ordinary Course of Business;
- suffered any damage, destruction or casualty loss to its tangible assets in excess of \$10,000, whether or not covered by insurance;
- (q) made any capital expenditures or commitments in excess of \$10,000 in the aggregate;
- (r) reduced or delayed any budgeted or planned capital expenditures;
- (s) made any change in any method of accounting or accounting policies;
- engaged in any promotional sale, discount, rebate, price reduction or other activity that has, or would reasonably be expected to have, the effect of accelerating to pre-Closing periods revenues that otherwise would be expected to occur in post-Closing periods;
- instituted or permitted any material change in the conduct of the Company's business, or any change in its method of purchase, sale, lease, management, marketing, promotion or operation;
- entered into, amended or terminated any material Contract, or any license or permit issued by any Governmental Authority;
- taken any other action or entered into any other transaction other than in the Ordinary Course of Business;
- entered into any other material transaction, whether or not in the Ordinary Course of Business, or materially changed any business practice;
- (y) entered into any Contract, or made any offer or commitment, to do any of the foregoing.

## 5.10 <u>Leased Real Property</u>.

- (a) The Company does not own, and has never owned, any real property. The Leased Real Property comprises all of the real property owned, occupied or otherwise used or intended to be used for, and necessary to operate, the business of the Company as currently conducted and as proposed to be conducted.
- (b) Section 5.10 of the Disclosure Schedule sets forth: (i) a list of the address and legal description of each parcel of Leased Real Property; (ii) the Lease with respect to each parcel of Leased Real Property; and (iii) the identity and contact information of each other party to each such Lease. Sellers have delivered to Buyer copies of each Lease with respect to the Leased Real Property, and each of the following documents in the possession or under the control of the Company, Sellers or their respective Affiliates with respect to all or any portion of the Leased Real Property: (1) environmental site assessments; (2) geotechnical, engineering or similar studies or assessments; (3) documents, information, reports, assessments, photographs or recordings concerning or relating to the condition of such Leased Real Property; and (4) title insurance policies, permits, warranties, guaranties, opinions, abstracts, plats and surveys.
- (c) Except as expressly set forth on <u>Section 5.10</u> of the Disclosure Schedule, with respect to each parcel of Leased Real Property:
  - the Lease applicable to such Leased Real Property is legal, valid, binding, enforceable and in full force and effect;
  - (ii) the transactions contemplated by this Agreement do not require the consent of any other party to any Lease applicable to such Leased Real Property, will not result in a breach of or default under such Lease, and will not otherwise cause such Lease to cease to be legal, valid, binding, enforceable and in full force and effect following the Closing on terms and conditions that are identical to those in effect prior to the Closing;
  - (iii) the Company's possession and quiet enjoyment of such Leased Real Property and under the Lease applicable thereto has not been disturbed, and there are no disputes with respect to such Leased Real Property or Leases;
  - (iv) neither the Company nor, to the Knowledge of Sellers, any other party to a Lease applicable to such Leased Real Property is in breach or default under such Lease, and no event has occurred or circumstance exists which, with the delivery of notice, the passage of time or both, could reasonably be expected to constitute such a breach or default, or permit the termination, modification or acceleration of rent under such Lease;
  - (v) the Company has not subleased, assigned, licensed or otherwise granted any Person the right to use, occupy or otherwise possess such Leased Real

- (vi) such Leased Real Property has direct vehicular access to a public street, or has vehicular access to a public street via a valid easement or license, and has parking for the employees, contractors and invitees of the Company, in each case sufficient for the operation of the Leased Real Property as currently operated by the Company;
- (vii) the utilities (including electricity, gas, water, telephone, internet, etc.), improvements, buildings and building systems (including electrical systems, plumbing, drainage, HVAC, communications, fire alarms and suppression, etc.) serving such Leased Real Property are, in each case, sufficient for the operation of the Leased Real Property as currently operated by the Company;
- (viii) no security deposit or portion thereof deposited with respect to a Lease applicable to such Leased Real Property has been applied in respect of a breach or default under such Lease which has not been re-deposited in full;
- (ix) the Company has not pledged, mortgaged or otherwise granted a Lien with respect to any portion of such Leased Real Property; and
- (x) there are no condemnation, rezoning or taking actions pending or, to the Knowledge of Sellers, threatened, with respect to any Leased Real Property.
- (d) The current use of the Leased Real Property does not violate any Laws, instrument of record or Contract affecting the Leased Real Property, and there is no violation of any covenant, condition, restriction, easement, agreement or order of any Governmental Authority having jurisdiction over the Leased Real Property that affects the use or occupancy thereof.
- 5.11 Assets. The Company has good and marketable title to, a valid license to use, or a valid leasehold interest in, free and clear of all Liens, all tangible assets used or useful in or necessary to conduct the Company's business (including but not limited to all materials, inventory, tools, machinery, equipment, vehicles, office equipment, computers, spare and replacement parts, and supplies). Each such tangible asset has been maintained in accordance with normal industry practice, is in good operating condition and repair (subject to normal wear and tear), and is suitable for the purposes for which it is currently used and, to the Knowledge of Sellers, each such tangible asset is free from defects.
  Section 5.11 of the Disclosure Schedule sets forth a list of the Company's material tangible assets.

#### 5.12 Tax Matters.

(a) The Company has filed on a timely basis all Tax Returns required to have been filed by the Company. All such Tax Returns are correct and complete in all respects. All Taxes required to have been paid by the Company (whether or not ever been made to the Company by a Governmental Authority in a jurisdiction where the Company does not file Tax Returns that the Company is or may be subject to taxation by that jurisdiction. The Company has not requested or obtained any extension of time within which to file any Tax Return, which Tax Return has not since been filed, except for automatic extensions of time to file which did not require the affirmative consent of the applicable Governmental Authority and are not currently due (as extended). There are no Liens on any of the assets of the Company that arose in connection with any failure (or alleged failure) to pay any Tax.

- (b) The Company has withheld or collected and timely paid over to the proper Governmental Authority all Taxes (including sales and use and employment Taxes) required by Law to have been withheld or collected and paid in connection with any amounts paid, distributed or owing to any Company employee, vendor, independent contractor, creditor, shareholder, or other third party, and all IRS Forms W-2, 1095C and 1099 (and similar state, local and foreign forms) required with respect thereto have been properly completed and timely filed. The Company has properly requested, received, and retained all necessary exemption certificates and other documentation supporting any claimed exemption with respect to sales or use Taxes.
- (c) There is no audit or other proceeding pending or threatened with regard to any Tax Liability or Tax Return of the Company.
- (d) The Company has not waived any statute of limitations in respect of Taxes or agreed to any extension of time with respect to a Tax assessment or deficiency.
- (e) No power of attorney has been granted by the Company with respect to any Tax matter relating to the Company.
- (f) The Company is not a party to any Contract, plan or arrangement requiring it to make payments to any Person that would be a "parachute payment" (within the meaning of Section 280G of the Code) as a result of any event connected with the acquisition of the Company by Buyer or any other transaction contemplated by this Agreement. The Company has not been a United States real property holding corporation within the meaning of Section 897(c)(2) of the Code at any time during the five-year period ending on the Effective Date.
- (g) The Company is not and has not been a party to any Tax allocation, Tax sharing or similar agreement or arrangement. Company has not been a member of an Affiliated Group filing a consolidated federal Tax Return. The Company does not have, and has not had a relationship to any other Person which would cause it to have, any Liability for Taxes owing by any other Person, including as a transferee or successor, by contract or otherwise.
- (h) The Company: (i) is not a party to any joint venture, partnership or other agreement or arrangement which is treated as a partnership for federal income

Tax purposes; and (ii) does not own any ownership interest in an entity that is treated as an entity disregarded as separate from its owner for federal Tax purposes, or as to which an election pursuant to Treasury Regulation Section 301.7701-3 has been made.

- The unpaid Taxes of the Company: (i) did not, as of the date of the Latest Balance Sheet, exceed the reserve for Tax Liability (rather than any reserve for deferred Taxes established to reflect timing differences between book and Tax income) set forth on the face of the Latest Balance Sheet (rather than in any notes thereto); and (ii) do not exceed that reserve as adjusted for the passage of time through the Closing in accordance with the past custom and practice of the Company in filing its Tax Returns. Since the date of the Latest Balance Sheet, the Company has not incurred any liability for Taxes arising from "extraordinary gains or losses," as that term is used in GAAP.
- (j) The Company will not be required to include any item of income in, or exclude any item of deduction from, taxable income for any taxable period (or portion thereof) ending after the Closing as a result of any:
  - change in method of accounting for a taxable period ending on or prior to the Effective Date;
  - (ii) "closing agreement" as described in Section 7121 of the Code (or any corresponding or similar provision of state, local or foreign income Tax law) executed on or prior to the Effective Date;
  - (iii) intercompany transaction or excess loss account described in Treasury Regulations under Section 1502 of the Code (or any corresponding or similar provision of state, local or foreign income Tax law) made on or prior to the Effective Date;
  - (iv) installment sale or open transaction disposition made on or prior to the Effective Date; or
  - (v) prepaid amount received on or prior to the Effective Date.
- (k) The Company has not distributed stock or securities of another Person, or has had its equity or securities distributed by another Person, in a transaction that was purported or intended to be governed in whole or in part by Section 355 of the Code or Section 361 of the Code.
- (I) The Company (and any predecessor of the Company) is and has at all times since its organization been taxed as a partnership within the meaning of the Code, and will continue to be taxed as a partnership within the meaning of the Code at all times up to the time of Closing.
- (m) In the past 5 years, the Company has not: (i) acquired assets from another corporation in a transaction in which the Company's Tax hasis for the acquired

102596957.7 0059466-00008

assets was determined, in whole or in part, by reference to the Tax basis of the acquired assets (or any other property) in the hands of the transferor; or (ii) acquired the stock of any corporation that is or was a qualified subchapter S subsidiary.

#### 5.13 Contracts.

- (a) Section 5.13(a) of the Disclosure Schedule sets forth an accurate and complete list of each of the following written Contracts, and a true, correct and complete written description of each of the following oral or other unwritten Contracts, to which the Company is a party or which affects the Company or its business, properties, assets or rights (each such Contract, a "Material Contract"):
  - Contracts to which any Seller, any Affiliate of the Company or any Seller, or any of their current or former officers or directors, is a party;
  - (ii) Contracts for the acquisition of any Person or any business unit or assets thereof, or the disposition of any assets of the Company, other than Contracts pursuant to which the applicable acquisition or disposition has been consummated and in respect to which the Company has not had any continuing obligations since;
  - (iii) Contracts granting any Person any preferential right to purchase, license, exercise or acquire any properties, assets or rights of the Company;
  - (iv) Contracts relating to any Indebtedness, or the making of any Ioans by the Company to any Person;
  - (v) Contracts relating to the ownership of or investment in any Person, including investments in joint ventures and minority equity investments;
  - (vi) any settlement, conciliation or similar Contract pursuant to which the Company will have any performance obligations after the Closing;
  - (vii) Contracts requiring capital expenditures after the date of this Agreement;
  - (viii) Contracts for the employment or engagement of any director, officer, employee, independent contractor or other Person on a full-time, parttime, consulting or other basis, including those (A) providing for the payment of any cash or other compensation or benefits upon the consummation of the transactions contemplated hereby or upon termination of employment or engagement, or (c) containing noncompete, nonsolicitiation, nondisclosure, confidentiality provisions or other restrictive covenants;
  - (ix) Contracts relating to the development, assignment, transfer, enforcement or licensing of Proprietary Rights, excluding licenses for unmodified "off the shelf", commercially available software applications with an

102596957.7 0059466-00008

- acquisition cost and/or aggregate annual license and maintenance fee of less than \$1,000;
- (x) Leases and all other Contracts relating to the lease, license, use or occupancy of real property, equipment, vehicles, other personal property or fixtures;
- (xi) Contracts restricting or purporting to restrict the Company, or any or any other Person for the benefit of the Company, from: (A) engaging in any line of business or competing with any Person; (B) engaging in any activity in any geographical area; or (C) soliciting any Person to enter into a business or employment relationship or entering into such a relationship with any Person;
- (xii) Contracts which require payment of consideration, or delivery of assets, goods, materials or services: (A) to the Company in an amount in excess of \$5,000 per year; or (B) by the Company;
- (xiii) Contracts that obligate the Company to comply with an exclusivity or "most favored nations" clause;
- (xiv) Contracts requiring the Company to purchase all or substantially all of its requirements of any product or service from any Person, that contain minimum purchase requirements provisions, or that require the Company to conduct business on an exclusive basis with any Person;
- (xv) Contracts with any Governmental Authority;
- (xvi) Contracts that are entered into outside of the Ordinary Course of Business or are otherwise material to the Company; and
- (xvii) Contracts committing the Company to enter into any of the foregoing.
- (b) Except as specifically disclosed on Section 5.13(b) of the Disclosure Schedule: (i) the Company, and, to the Knowledge of Sellers, the other party thereto, have performed all obligations required to be performed by such Person under each of the Material Contracts; (ii) there is no breach of or default under such Contract or any event which, upon giving of notice or lapse of time or both, would constitute a breach or default under such Material Contract by the Company or, to the Knowledge of Sellers, by the other party thereto; (iii) there is no anticipated breach by any party to any Material Contract; (iv) the Company, and, to the Knowledge of Sellers, the other party to such Material Contract, has not assigned, delegated or otherwise transferred to any Person any of its rights, title or interest under any Material Contract; (v) to the Knowledge of Sellers each Material Contract is legal, valid, binding, enforceable and in full force and effect, and will continue as such following the consummation of the transactions contemplated

- the application of specific performance and other equitable principles); and (vi) the transactions contemplated by this Agreement shall not impair the rights of the Company under, or cause a breach or default of, any Material Contract or cause any additional fees or royalties to be due thereunder.
- (c) Buyer has been supplied with a true and correct copy of all written contracts, and a true, correct and complete written description of all oral and other unwritten Contracts, which are referred to on <u>Section 5.13(a)</u> of the Disclosure Schedule, together with all amendments, exhibits, attachments, waivers or other changes thereto.
- 5.14 Proprietary Rights. The Company owns or has the valid right to use all Proprietary Rights necessary for the operation of the Company's business as currently conducted and as proposed to be conducted. Section 5.14 of the Disclosure Schedule sets forth a complete and correct list of: (a) all patented or registered Proprietary Rights and pending patent applications or other applications for registration of Proprietary Rights; (b) all trade names used by the Company; (c) all licenses, consents, Contracts or arrangements to which the Company is a party, either as licensee or licensor, concerning Proprietary Rights or part thereof; and (d) all Company Marks, and any other material unregistered trademarks and copyrights that are either owned by the Company or used or held by the Company or any of its Affiliates in the conduct of the Company's business (pursuant to license agreement or otherwise, in such case identifying the owner and title of the agreement pursuant to which the Proprietary Right is used). Such Proprietary Rights comprise all of the intellectual property rights necessary or desirable for the conduct of the Company's business as currently conducted and as currently proposed to be conducted. Except as set forth in Section 5.14 of the Disclosure Schedule:
  - (a) To the Knowledge of Sellers the Company owns and possesses exclusively all right, title and interest in, to and under the Proprietary Rights, free and clear of all Liens, and no claim by any third party contesting the validity, enforceability, use, registration or ownership of any of the Proprietary Rights has been made, is currently outstanding or, to the Knowledge of Sellers, is threatened;
  - (b) The Company has not transferred ownership of or granted any license of or other right to use or authorized the retention of any rights to use any Proprietary Rights to any other Person;
  - (c) No loss or expiration of any part of Proprietary Rights has occurred, is pending or, to the Knowledge of Sellers, threatened or reasonably foreseeable;
  - (d) Neither the Company nor any Seller has received any notice of, and to the Knowledge of Sellers, there are no facts which indicate a likelihood of, any infringement or misappropriation by, or conflict with, any third party with respect to the Proprietary Rights, including any demand or request that the Company license rights from a third party;

- (e) None of the Proprietary Rights nor the Company has infringed, misappropriated or otherwise come into conflict with any rights of any third parties, and Sellers have no Knowledge of any infringement, misappropriation or conflict which will occur as a result of the continued operation of the Company's business as currently conducted or as currently proposed to be conducted;
- (f) To the Knowledge of Sellers neither the Company, nor Sellers, nor their respective Affiliates has disclosed any of the Company's trade secrets or Confidential Information to any third party other than pursuant to a written, valid and binding confidentiality agreement in favor of the Company; and
- (g) The Company has executed all appropriate Contracts with current and past employees, contractors and agents to assign to the Company all of their right, title and interest in any Proprietary Rights and has furnished Buyer with true and correct copies of all such Contracts.
- 5.15 <u>Litigation</u>. Except as set forth on <u>Section 5.15</u> of the Disclosure Schedule, there are no and, during the five (5) years preceding the Effective Date, there have not been any, actions, suits, proceedings, orders or investigations pending or, to the Knowledge of Sellers, threatened against or affecting the Company, its business or assets, or the Membership Interests, whether at law or in equity, or before or by any Governmental Authority (including any actions, suits, proceedings, orders or investigations with respect to the transactions contemplated by this Agreement, or any condemnation, expropriation or other proceeding in eminent domain affecting any real property or any portion thereof), and there is no reasonable basis for any of the foregoing. The Company is not subject to any grievance or arbitration proceedings under collective bargaining agreements or otherwise, or any governmental investigations or inquiries. Neither the Company, nor Sellers, nor any of their respective Affiliates have received any opinion or memorandum or advice from legal counsel to the effect that the Company is exposed, from a legal standpoint, to any material Liabilities.
- 5.16 <u>Brokerage</u>. There are no claims, and the Company has no Liability, for brokerage commissions, finders' fees or similar compensation in connection with the transactions contemplated by this Agreement based on any arrangement or agreement made by or on behalf of the Company, Sellers or their respective Affiliates.

#### 5.17 Employees.

(a) Section 5.17(a) of the Disclosure Schedule lists all of the employees of the Company (the "Company Employees"), together with each Company Employee's title, date of hire, leave status, current salary, bonus or incentive compensation eligibility, and any accrued paid leave (including, without limitation, under any policy of the Company relating to vacation, personal time off or otherwise). Except as set forth on Section 5.17(a) of the Disclosure Schedule: (i) the Company does not have a written employment, consulting or similar Contract with any manager, Company Employee, independent contractor, agent, consultant or professional adviser of the Company; (ii) there are no severance payments,

change-in-control payments or similar payments which are or could become payable by the Company or any other Person to any manager, Company Employee, independent contractor, agent, consultant or professional adviser of the Company under the terms of any oral or written Contract or any Law, custom, trade or practice; (iii) there are no other material Contracts, oral or written, between the Company and any such person relating to such person's employment or engagement; and (iv) to the Knowledge of Sellers, no officer or management level or technical employee or independent contractor of the Company has any plans to terminate his, her or their employment, engagement or relationship with the Company.

- (b) Except as set forth on Section 5.17(b) of the Disclosure Schedule:
  - the Company is not a party to any collective bargaining agreement or any other labor-related agreements with any labor union or bargaining unit applicable to employees of the Company, nor is any such agreement currently being negotiated;
  - (ii) no labor organization or group of employees has filed any representation petition or made any written or oral demand for recognition;
  - (iii) to the Knowledge of Sellers, no union organizing or decertification efforts are pending or threatened, and no other question concerning representation exists;
  - (iv) no labor strike, work stoppage, slowdown, or other material labor dispute has occurred at any time, and none is pending or, to the Knowledge of Sellers, threatened;
  - (v) there are no charges against the Company pending or, to the Knowledge of Sellers, threatened in writing before the Equal Employment Opportunity Commission or any state, local or foreign agency responsible for the prevention of unlawful employment practices; and
  - (vi) there is no unfair labor practice charge or complaint against the Company pending or, to the Knowledge of Sellers, threatened before the National Labor Relations Board or any comparable state, local or foreign agency.
- (c) With respect to the transaction contemplated by this Agreement, any notice required under any Law has been given. Within the past two (2) years, the Company has not implemented any plant closing or layoff of employees that could implicate the Worker Adjustment and Retraining Notification Act of 1988, as amended, or any similar foreign, state, or local law, regulation, or ordinance.

- (d) Section 5.17(d) of the Disclosure Schedule sets forth a list of all Persons who are performing or have performed services for the Company as an independent contractor, leased employee, temporary employee or consultant and who are not treated by the Company for all Tax purposes as regular employees.
- 5.18 Product and Service Warranties. Except as expressly set forth in the Material Contracts listed on Section 5.13(a) of the Disclosure Schedule, the Company has not made any express warranties or guarantees with respect to the products provided or services rendered by or on behalf of the Company in connection with its business. To the Knowledge of Sellers each product provided and service rendered by or on behalf of the Company has been in conformity in all material respects with all applicable contractual commitments and all express, implied or statutory warranties, and the Company does not have any material Liability for replacement, repair or re-performance thereof or other damages in connection therewith, subject only to the reserve for warranty claims (if any) set forth on the Latest Balance Sheet and included as a Closing Current Liability in the calculation of the final Adjustment Amount.

## 5.19 Employee Benefit Plans.

- (a) Section 5.19 of the Disclosure Schedule contains a true and complete list of: (i) each deferred compensation, incentive compensation, and equity compensation plan or agreement; (ii) each severance or termination pay, medical, surgical, hospitalization, life insurance and other "welfare" plan, fund or program (within the meaning of Section 3(1) of the Employee Retirement Income Security Act of 1974, as amended ("ERISA")); (iii) each profit-sharing or other "pension" plan, fund or program (within the meaning of Section 3(2) of ERISA); (iv) each employment agreement providing deferred compensation, severance or termination benefits; and (v) each other employee benefit plan, fund, program or agreement; in each case as to items (i) through (v) foregoing, that is sponsored, maintained or contributed to by the Company and, solely with respect to clause (iii), by any trade or business, whether or not incorporated, that together with the Company would be deemed a "single employer" within the meaning of Section 4001(b) of ERISA (an "ERISA Affiliate"), or to which the Company or an ERISA Affiliate is party, that, in each such case, is for the benefit of any employee or former employee of the Company (each, a "Company Plan"), or with respect to a Company Plan described in clause (iii), maintained or which the Company or an ERISA Affiliate has contributed within the past five (5) years.
- (b) Each Company Plan (and each related trust, insurance contract or fund) has been maintained, funded and administered in accordance with its terms and in compliance with all applicable Laws, including, without limitation, ERISA, the Health Insurance Portability and Accountability Act, the Patient Protection and Affordable Care Act, the Pension Protection Act of 2006 and the Code. With respect to each Company Plan, all required payments, premiums, contributions, distributions or reimbursements for all periods ending prior to or as of the Effective Date have been made or properly accrued. Each Company Plan that is intended to be a "qualified plan" under Section 401(a) of the Code has received a

- favorable determination letter from the IRS dated in 2007 or later covering all applicable requirements of the Code, other than those for which the remedial amendment period under Section 401(b) of the Code has not expired, and nothing has occurred that could adversely affect the qualification of such Company Plan.
- (c) With respect to each Company Plan, the Company has provided Buyer with true and complete copies of the following (as applicable): (i) the annual report on IRS Form 5500 as filed for the last three (3) years (including all schedules and attachments), if required; (ii) the summary plan description and any summaries of material modification; (iii) the plan document and all amendments thereto; (iv) all trust agreements, insurance contracts, investment policies and similar instruments; (v) any determination letter issued by the IRS; and (vi) nondiscrimination and elective contribution testing and supporting materials for the most recent plan year for each Company Plan required to undertake such testing.
- (d) No Company Plan provides post-employment or post-termination medical or life or other welfare or welfare-type benefits, other than as required by COBRA. The Company and the ERISA Affiliates have complied and are in compliance with the requirements of COBRA. No Company Plan that is a medical or health plan: (i) is being administered as a "grandfathered health plan" within the meaning of DOL Regulation §2590.715-1251; or (ii) is a self-insured plan whether or not covered by a "stop-loss" insurance policy.
- (e) Neither the Company nor any ERISA Affiliate maintains, sponsors, contributes to, or has any Liability under or with respect to any: (i) "defined benefit plan" (as defined in Section 3(35) of ERISA); (ii) "multiemployer plan" (as defined in Section 3(37) or 4001(a)(3) of ERISA); (iii) "multiple employer plan" (as defined in Section 210 of ERISA or Section 413(c) of the Code); or (iv) "multiple employer welfare arrangement" (as defined in Section 3(40) of ERISA). The Company does not have any Liability as a consequence of at any time being considered a single employer under Section 414 of the Code with any other Person. No Company Plan is subject to any tax under Section 512 of the Code.
- (f) With respect to each Company Plan: (i) there have been no prohibited transactions as defined in Section 406 of ERISA or Section 4975 of the Code; (ii) no fiduciary (as defined in Section 3(21) of ERISA) has any Liability for breach of fiduciary duty or any other failure to act or comply in connection with the administration or investment of the assets of such plans; and (iii) no actions, audits, investigations, proceedings, suits or claims (other than routine claims for benefits) are pending or threatened, and there are no facts that would give rise to or could reasonably be expected to give rise to any such actions, audits, investigations, proceedings, suits or claims.
- (g) Each Company Plan that contains, is or forms part of a "non-qualified deferred compensation plan" within the meaning of Section 409A of the Code complies in form and has been operated in compliance with the requirements of Section 409A of the Code and the guidance issued thereunder. The Company has no obligation

- to gross-up, indemnify or otherwise reimburse any Person for any Tax incurred by such Person pursuant to any applicable federal, state or local Law related to the collection or payment of Taxes.
- (h) The Company has, for purposes of each Company Plan, correctly classified those individuals performing services for the Company as common law employees, leased employees, independent contractors or agents.
- (i) The transactions contemplated by this Agreement will not cause the acceleration of vesting in, or payment of, any benefits under any Company Plan or any other agreement and will not otherwise accelerate or increase any Liability under any Company Plan.
- Insurance. The Company is insured by policies of insurance maintained by the Company in amounts and scope of coverage as set forth in <a href="Section 5.20">Section 5.20</a> of the Disclosure Schedule. Each insurance policy of the Company is in full force and effect and all premiums are currently paid in accordance with the terms of such policy or accrued. Neither the Company, nor Sellers, nor their respective Affiliates, has received any notice that any such insurance policy will be cancelled or will not be renewed, nor has the Company, nor Sellers, nor their respective Affiliates received any written notice that cancellation or non-renewal is threatened nor any written notice that any material modification of the terms of policy of insurance will be or is threatened to be required as a condition of renewal. The Company has at all times maintained such insurance coverage required pursuant to then-applicable Law or pursuant to Contract to which the Company is or was then a party or by which the Company or any of its assets are or were then bound or affected. The Company does not have any self-insurance or co-insurance programs.
- 5.21 <u>Compliance with Laws; Permits.</u> Except as set forth on <u>Section 5.21</u> of the Disclosure Schedule:
  - (a) The Company has complied and is in compliance with all applicable Laws, and no notices have been received by, and no claims have been filed against, the Company alleging a violation of any such Laws;
  - (b) The Company has held, and currently holds, all permits, licenses, certificates, accreditations or other authorizations of Governmental Authorities required for the conduct of its business (including, without limitation, those required under Environmental and Safety Laws), and <a href="Section 5.21">Section 5.21</a> of the Disclosure Schedule sets forth a list of all of such permits, licenses, certificates, accreditations and other authorizations. The Company is in compliance with all terms and conditions of any such required permits, licenses, accreditations and authorizations. The Company has not received any notice or citation for noncompliance with any of its permits, licenses, accreditations and authorizations, and there exists no condition, situation or circumstance, nor has there existed such a condition, situation or circumstance, which, after notice or lapse of time, or both, would constitute noncompliance with or give rise to future material Liability

- with regard to any of the foregoing permits, licenses, accreditations and authorizations.
- (c) Each employee, contractor and agent of the Company (and each employee and subcontractor of such contractors or agents) has held, and currently holds, all permits, licenses, certificates, accreditations or other authorizations of Governmental Authorities required for the performance of any services by such employee, contractor or agent on behalf of the Company. To the Knowledge of Sellers, each such employee, contractor and agent of the Company (and each employee or subcontractor of such contractors or agents) is in compliance with all terms and conditions of any such required permits, licenses, accreditations and authorizations.
- (d) No officer, manager, employee, consultant, advisor, agent or representative of the Company has been authorized to make or receive, and has not made or received, directly or indirectly, any bribe, kickback payment or other illegal payment, whether in cash or in kind, at any time.
- 5.22 <u>Environmental Matters</u>. Except as set forth on <u>Section 5.22</u> of the Disclosure Schedule:
  - (a) The Company and its predecessors have complied and are in compliance with all Environmental and Safety Laws;
  - (b) Without limiting the generality of the foregoing, the Company and its Affiliates have obtained and complied with, and are in compliance with, all permits, licenses, accreditations and authorizations that are required pursuant to Environmental and Safety Laws for the occupation of their facilities and the operation of the Company's business; and a list of all such permits, licenses, accreditations and authorizations is set forth on <u>Section 5.21</u> of the Disclosure Schedule.
  - (c) Neither the Company nor any of its predecessors or Affiliates has received any oral or written notice, report or other information regarding any alleged or actual violation of, or Liability under, Environmental and Safety Laws, including any Remediation, investigatory, remedial or corrective obligations arising under Environmental and Safety Laws, and no basis exists for any such Liability;
  - (d) None of the following exists or has existed at any property or facility owned, leased, licensed or operated by the Company: (i) underground storage tanks; (ii) above-ground storage tanks, (iii) asbestos-containing material in any form or condition, (iv) materials or equipment containing polychlorinated biphenyls; or (v) landfills, surface impoundments, underground injection wells or disposal areas.
  - (e) Neither the Company, nor any of its predecessors or Affiliates with respect to which the Company may be deemed a successor or have any Liability under Environmental and Safety Laws, has treated, stored, disposed of, arranged for or

to, any Hazardous Materials, or owned, leased, licensed or operated any property or facility, and no such property or facility owned, leased, licensed or operated by the Company or its predecessors is contaminated by any such Hazardous Materials or substance, in a manner that has given or would give rise to any Liabilities or investigative, corrective or remedial obligations pursuant to any applicable Environmental and Safety Laws.

- (f) No facts, events or conditions relating to the past or present facilities, properties or operations of the Company, or its predecessors or Affiliates, will prevent, hinder or limit the Company's continued compliance with Environmental and Safety Laws, give rise to any investigatory, Remediation or corrective obligations pursuant to Environmental and Safety Laws, or give rise to any other Liabilities (whether accrued, absolute, contingent, unliquidated or otherwise) pursuant to Environmental and Safety Laws, including any relating to onsite or offsite releases or threatened releases of Hazardous Materials, substances or wastes, personal injury, property damage or natural resources damage.
- (g) Neither this Agreement nor the consummation of the transactions contemplated hereby will result in any obligations for site investigation or cleanup, or notification to or consent of Governmental Authorities or third parties, pursuant to any of the so-called "transaction-triggered" or "property transfer" Environmental and Safety Laws.
- (h) Neither the Company, nor any of its predecessors or Affiliates with respect to which the Company may be deemed a successor or have any Liability under Environmental and Safety Laws, has, either expressly or by operation of law, assumed or undertaken any Liability, including any obligation for Remediation, of any other Person relating to Environmental and Safety Laws.
- (i) Sellers have furnished to Buyer all environmental audits, reports, correspondence to or from Governmental Authorities and other material environmental documents relating to the past or current operations or facilities of the Company, and those of its predecessors or Affiliates with respect to which the Company may be deemed a successor or have any Liability under Environmental and Safety Laws, which are in the possession or under the reasonable control of the Company, Sellers or their respective Affiliates.
- 5.23 <u>Customers and Suppliers.</u> <u>Section 5.23</u> of the Disclosure Schedule accurately sets forth a list of the top ten (10) (a) customers of the Company (each, a "<u>Material Customer</u>") and (b) suppliers or vendors of the Company (each, a "<u>Material Supplier</u>"), determined and organized in descending order by volume of sales or purchases (as applicable) for the fiscal years ended December 31, 2018 and the interim period ended as of date of the Latest Balance Sheet. Except as set forth on <u>Section 5.23</u> of the Disclosure Schedule, neither the Company, nor Sellers, nor their respective Affiliates has received any indication from any Material Customer or any Material Supplier to the effect, and neither the Company, nor Sellers, nor their respective Affiliates has any reason to believe, that such Material Customer or Material Supplier will terminate or not renew any Contract

- with the Company, or stop, or materially decrease the rate of, buying or supplying products or services from or to the Company.
- 5.24 Affiliate Transactions. Except as set forth on Section 5.24 of the Disclosure Schedule:
  - (a) No Seller, nor any officer, manager, employee or Affiliate of the Company or any Seller, nor any individual related by blood, marriage or adoption to any such individuals, nor any other Person in which any such Person holds a legal or beneficial interest: (i) owns any direct or indirect legal or beneficial interest in the Company; (ii) is a party to any Contract, transaction or arrangement to which the Company is also a party; or (iii) has any direct or indirect legal or beneficial interest in any property, asset or right owned or used by the Company (including any Proprietary Rights).
  - (b) Section 5.24 of the Disclosure Schedule sets forth each individual employed or engaged by the Company that is related by blood, marriage or adoption to any Seller or any officer, manager, or Affiliate of the Company or any Seller.
  - (c) Section 5.24 of the Disclosure Schedule sets forth a description of each of the services provided to the Company by any Seller or any of their respective Affiliates and the cost and other material terms thereof.
- 5.25 <u>Bank Accounts; Powers of Attorney</u>. <u>Section 5.25</u> of the Disclosure Schedule sets forth a true, correct and complete list of the names and locations of all banks and other financial institutions at which the Company maintains an account or safe deposit box, the names of all Persons authorized to withdraw therefrom or that have access thereto and the names of all Persons holding powers of attorney from the Company as of the date of this Agreement.
- 5.26 <u>Disclosures</u>. Neither this Agreement, nor any of the schedules, exhibits or certificates attached hereto or delivered in connection herewith contain any untrue statement of a material fact or omit a material fact necessary to make the statements contained herein or therein, in light of the circumstances in which they were made, not misleading. There is no event, circumstance or other fact which has not been disclosed to Buyer in this Agreement and the schedules attached hereto which has had or would reasonably be expected to have a Material Adverse Effect.

# **ARTICLE VI**

### REPRESENTATIONS AND WARRANTIES OF BUYER

As an inducement to Sellers to enter into this Agreement and consummate the transactions contemplated hereby, Buyer hereby represents and warrants to Sellers that all of the following representations and warranties in this <u>ARTICLE VI</u> are true and correct as of the date of this Agreement and as of the Closing:

- 6.1 <u>Organization and Corporate Power</u>. Buyer is a corporation duly formed and validly existing under the laws of the State of Nevada, with full corporate power and authority to enter into this Agreement and to perform its obligations hereunder.
- Authorization. The execution, delivery and performance of this Agreement and the other agreements contemplated hereby to be executed and delivered by Buyer and the consummation of the transactions contemplated hereby and thereby have been duly and validly authorized by all requisite corporate action on the part of Buyer, and at the Closing no other corporate proceedings on the part of Buyer will be necessary to authorize the execution, delivery or performance of this Agreement or the other agreements contemplated hereby. This Agreement, and the other agreements contemplated hereby to be executed and delivered by Buyer, constitute valid and binding obligations of Buyer, enforceable against Buyer in accordance with their respective terms, except to the extent such enforceability may be limited by bankruptcy, insolvency or similar laws, or by general principles of equity, applicable to creditors' rights generally.
- Governmental Authorities and Consents. No consent, approval or authorization of any Governmental Authority or other third party is required to be obtained by Buyer in connection with the execution and delivery of this Agreement or the consummation of the transactions contemplated hereby.
- 6.4 <u>Brokerage</u>. There are no claims for brokerage commissions, finders' fees or similar compensation in connection with the transactions contemplated by this Agreement based on any arrangement or agreement made by or on behalf of Buyer.
- 6.5 <u>Litigation</u>. There are no actions, suits, proceedings, orders or investigations pending or, to the knowledge of Buyer, threatened against or affecting Buyer at law or in equity, or before or by any Governmental Authority which would adversely affect Buyer's performance under this Agreement or the consummation of the transactions contemplated hereby.
- 6.6 Acquisition for Investment. Buyer is not acquiring the Membership Interests with a view to, or for sale in connection with, any distribution thereof within the meaning of the Securities Act.
- 6.7 <u>Valid Issuance</u>. The shares of Buyer Stock issued as Closing Stock Consideration pursuant to this Agreement will be, as of the Closing, validly issued, fully-paid and non-assessable share of Buyer Stock.

#### ARTICLE VII

#### COVENANTS

7.1 Expenses. Each Party hereto shall pay all of its own costs and expenses (including attorneys', accountants' and investment bankers' fees and other out-of-pocket expenses) in connection with the negotiation and execution of this Agreement, the performance of

provided, that Sellers shall pay all such costs and expenses incurred by or billed to the Company (other than any such costs and expenses incurred by Profire). Without limiting the foregoing, each Party shall pay its own expenses incurred in connection with its efforts to satisfy the conditions to the other Party's obligation to consummate the transactions contemplated hereby.

- Further Transfers; Transition Assistance. Each Party agrees that from time to time after the Closing, it will execute and deliver (or cause its Affiliates to execute and deliver) such further instruments, and take (or cause its Affiliates to take) such other action, as may be reasonably necessary to consummate the transactions contemplated by this Agreement and vest Buyer with full title to all Membership Interests and all assets, properties, privileges, rights, approvals and immunities of the Company and its business, or to effect the other purposes of this Agreement. Each Seller agrees that subsequent to the Closing it shall (and shall cause its Affiliates to) refer all customer inquiries with respect to the Company and its business to the Company. Each of the Parties will use its best efforts to take all actions and to do all things necessary, proper or advisable in order to consummate and make effective the transactions contemplated by this Agreement (including satisfaction, but not waiver, of the covenants and conditions set forth in this Agreement).
- 7.3 Confidentiality. Each Seller shall (and shall cause its Affiliates to) treat as strictly confidential and hold as such all of the Confidential Information, refrain from using or disclosing any of the Confidential Information except pursuant to this Agreement, and deliver promptly to Buyer or destroy, at the request and option of Buyer, all tangible embodiments (and all tangible or electronic copies) of the Confidential Information that are in his, her or its possession. In the event that any Seller or their Affiliates is required pursuant to a subpoena or a demand for the production of information or documents in any legal proceeding, interrogatory, subpoena, civil investigative demand, or similar process to disclose any Confidential Information, such Seller: (i) shall promptly notify Buyer of such requirement; (ii) shall cooperate with Buyer and the Company to contest such disclosure or seek an appropriate protective order; and (iii) only thereafter, may disclose only so much of such Confidential Information as such Seller or its Affiliates, in the good faith opinion of their legal counsel, are legally compelled to disclose pursuant thereto or else stand liable for contempt; provided, however, that the disclosing Seller and its Affiliates shall use their best efforts to obtain, at the request and expense of Buyer, an order or other assurance that confidential treatment will be accorded to any such Confidential Information disclosed pursuant to this Section 7.3. The foregoing provisions shall not apply to any Confidential Information that is generally available to the public immediately prior to the time of disclosure, unless such Confidential Information is so available due to the actions of Sellers or their Affiliates in violation of the terms of this Agreement or any other Contract with respect to such Confidential Information.
- 7.4 Release and Waiver. As a material inducement to Buyer to enter into this Agreement, effective as of the Closing, each Seller, on behalf of itself and its Affiliates, agrees not to sue and fully releases and discharges the Buyer Indemnified Parties with respect to and from any and all Losses. Liens, Liabilities (including employment contracts), covenants.

montant and an execution entries (mondaing ampropriate contractor) constraint

102596957.7 0059466-00008

or proceedings of whatever kind or nature in law, equity or otherwise, whether now known or unknown, suspected or alleged, contingent or absolute, all of which such Seller or its Affiliates now owns or holds or has at any time owned or held against any of the Buyer Indemnified Parties, including those arising out of or relating in any way to: (a) matters or circumstances occurring or existing prior to or as of the Closing; (b) the employment or engagement of any Seller with the Company prior to or as of the Closing, including the commencement, renewal, termination and/or separation of such employment at any time or for any reason; and (c) the purchase, acquisition, ownership, sale, transfer or other disposition of the Membership Interests; provided, that nothing in this Section 7.4 will be deemed to constitute a release or waiver by any Seller of any right to enforce any obligation of Buyer under this Agreement. It is the intention of each Seller that such release be effective as a bar to each and every claim, demand and cause of action hereinabove specified. In furtherance of this intention, each Seller hereby expressly waives any and all rights and benefits conferred upon such Seller or its Affiliates by the provisions of Law and expressly consents that this release will be given full force and effect according to each and all of its express terms and provisions, including those related to unknown and unsuspected claims, demands and causes of action, if any.

#### 7.5 <u>Non-Competition and Non-Solicitation</u>.

- (a) For the period commencing with Closing and ending on the date three (3) years after the Effective Date, each Seller shall not, and shall cause its Affiliates not to, for any reason whatsoever, directly or indirectly, for itself, himself or herself, or on behalf of or in conjunction with any other Person, take any of the following actions or engage in any of the following conduct anywhere in the United States or Canada:
  - (i) engage as an equity holder, officer, director, owner, partner, joint venturer, or in a managerial capacity, whether as an employee, independent contractor, consultant or advisor, in any business selling any products or the Company, Buyer or their Affiliates; provided, that the restrictions set forth in this Section 7.5(a)(i) shall not prohibit Dustin Baker from owning or holding a passive, minority equity interest in Last Arrow Manufacturing, so long as Dustin Baker does not actively participate in the management, direction or activities of Last Arrow Manufacturing and complies with all of his obligations set forth in Section 7.3 and this Section 7.5;
  - (ii) sell, offer, design, market, install, service or provide products or services in competition with those currently sold, offered, designed, marketed, serviced or contemplated by the Company, Buyer or their Affiliates, or call upon for the purpose of soliciting or selling, offering or providing products or services in competition with those currently offered or contemplated by the Company, Buyer or their Affiliates, to any Person who is at that time, or has been within the five (5) year period immediately

- preceding the Closing, a customer or prospective customer of the Company, Buyer or their Affiliates; or
- (iii) (A) induce or attempt to induce any employee or independent contractor of the Company, Buyer or their Affiliates, to terminate or diminish its employment or engagement with such entity, or in any way interfere with the relationship between the Company, Buyer or their Affiliates and any such employee or independent contractor thereof, (B) hire or engage any Person who was an employee or independent contractor of the Company, Buyer or their Affiliates, at any time during the one (1) year period immediately preceding the Closing or thereafter, or (C) induce or attempt to induce any customer, supplier, vendor, service provider, employee, independent contractor, licensee, licensor, lessor or other business relation of the Company, Buyer or their Affiliates, to cease or reduce its business with the Company, Buyer or their Affiliates or their respective businesses, or in any way interfere with the relationship between the Company, Buyer or their Affiliates and any such customer, supplier, vendor, service provider, employee, independent contractor, licensee, licensor, lessor or other business relation (including making any negative statements or communications about the Company, Buyer or their Affiliates or their respective businesses).

The Parties intend that the covenants contained in this <u>Section 7.5(a)</u> shall be deemed to be a series of separate covenants, one for each state of the United States and each province of Canada and, except for geographic coverage, each such separate covenant shall be identical in terms to the covenant contained in this Section 7.5(a).

- (b) Each Party acknowledges and agrees that the foregoing covenants in this <u>Section 7.5</u> impose a reasonable restraint on Sellers and their Affiliates in light of the activities and business of Buyer and the Company on the date of the execution of this Agreement.
- All of the covenants in this <u>Section 7.5</u> shall be construed as an agreement independent of any other provision in this Agreement, and the existence of any claim or cause of action of any Seller Indemnified Party against Buyer or the Company, whether predicated on this Agreement or otherwise, shall not constitute a defense to the enforcement by Buyer of such covenants. It is understood by the Parties that the covenants contained in this <u>Section 7.5</u> are essential elements of this Agreement and that, but for the agreement of Sellers and their Affiliates to comply with such covenants, Buyer would not have agreed to enter into this Agreement. Each Seller and Buyer have independently consulted with their respective counsel and have been advised concerning the reasonableness and propriety of such covenants with specific regard to the nature of the Company's business and the business conducted by Buyer. Each Seller hereby agrees that all covenants contained in this <u>Section 7.5</u> are material, reasonable and valid, and voluntarily, knowingly and irrevocably waives all defenses to the strict

43

102596957.7 0059466-00008

- enforcement hereof by Buyer. The covenants in this <u>Section 7.5</u> are severable and separate, and the unenforceability of any specific covenant in any jurisdiction shall not affect the enforceability of such covenant in any other jurisdiction, or the provisions of any other covenant. Moreover, in the event any court of competent jurisdiction shall determine that the scope, time or territorial restrictions set forth are unreasonable o unenforceable as written, then it is the intention of the Parties that such restrictions be enforced to the fullest extent which the court deems reasonable and enforceable, and the Agreement shall thereby be reformed. The covenants contained in this <u>Section 7.5</u> shall not be affected by any breach of any other provision hereof by any Party.
- (d) Because of the difficulty of measuring economic losses to Buyer as a result of a breach of the foregoing covenants in this <u>Section 7.5</u>, and because of the immediate and irreparable damage that could be caused to Buyer for which Buyer would have no other adequate remedy, each Seller agrees that the foregoing covenant may be enforced by Buyer in the event of any actual or threatened breach by any Seller or its Affiliates, by an injunction, restraining order, specific performance and other equitable remedies, without the requirement of posting a bond or other security, and in addition to, but not in lieu of, any other available remedies pursuant to this Agreement, by statute or at law (including monetary damages).
- Public Announcements. Buyer and the Seller Representative shall communicate with each other and cooperate with each other prior to any public disclosure of the transactions contemplated by this Agreement. Each Party agrees not to (and agrees to cause its agents and representatives not to) issue any press release or make any other public announcement relating to this Agreement or the transactions contemplated by this Agreement without the prior written approval of Buyer and the Seller Representative; provided, that Buyer and its Affiliates may disclose this Agreement and the transactions contemplated hereby to the extent required by applicable Laws (including Securities Laws) and The NASDAQ Stock Market rules.
- 7.7 Lock-Up. Each Seller covenants and agrees, for a period of twelve (12) months immediately following the Effective Date (the "Lock Up Period"), not to, directly or indirectly, sell, offer, transfer, assign, exchange, dispose of or otherwise distribute, or grant any legal or beneficial interest in, grant any option or right with respect to, any shares of Buyer Stock received pursuant to this Agreement, or any other securities received in any stock split, stock dividend, conversion, exchange or recapitalization of such Buyer Stock. Sellers' acknowledge and agree that each certificate issued with respect to any shares of Buyer Stock received pursuant to this Agreement will bear a restrictive legend and stop-transfer instructions in accordance with the provisions of this Section 7.7 and the other restrictions on transfer imposed by applicable Laws. Following the end of the Lock Up Period, upon the request of Sellers, Buyer will reasonably cooperate with Sellers to remove from any certificate issued with respect to any shares of Buyer Stock received pursuant to this Agreement any restrictive legend evidencing the restrictions imposed pursuant to this Section 7.7.

#### **ARTICLE VIII**

#### TAX MATTERS

- 8.1 Tax Indemnification. Each Seller shall jointly and severally, up to such Seller's Pro Rata Share of such Losses, indemnify the Buyer Indemnified Parties and hold them harmless from and against any Losses attributable to: (a) all Taxes (or the non-payment thereof) of the Company for all taxable periods ending on or before the Effective Date and the portion through the Effective Date for any taxable period that includes (but does not end on) the Effective Date ("Pre-Closing Tax Period"), including, without limitation, any Tax imposed under Section 1374 of the Code; (b) all Taxes of any member of an affiliated, consolidated, combined or unitary group of which the Company or any of its predecessors is or was a member on or prior to the Closing, including pursuant to Treasury Regulation Section 1.1502-6 or any analogous or similar state, local, or foreign Law; and (c) any and all Taxes of any Person (other than the Company) imposed on the Company as a transferee or successor, by contract or pursuant to any Law, which Taxes relate to an event or transaction occurring before the Closing.
- 8.2 Straddle Period. In the case of any taxable period that includes (but does not end on) the Effective Date (a "Straddle Period"), the amount of any Taxes for the Pre-Closing Tax Period that are based on or measured by income or receipts of the Company or that are imposed on other activities that can be assigned to a specific date shall be determined based on an interim closing of the books as of the Closing (and for such purpose, the taxable period of any partnership or other pass-through entity in which the Company holds a beneficial interest shall be deemed to terminate at such time), and the amount of other Taxes of the Company for a Straddle Period that relates to the Pre-Closing Tax Period shall be deemed to be the amount of such Tax for the entire taxable period multiplied by a fraction, the numerator of which is the number of days in the taxable period ending on the Effective Date, and the denominator of which is the number of days in such Straddle Period.
- Responsibility for Filing Pre-Closing Tax Returns. Sellers shall at their cost prepare or cause to be prepared, and timely file or cause to be timely filed, all Tax Returns of the Company with respect to any Pre-Closing Tax Period. Any such Tax Return shall be prepared in a manner consistent with prior practices of the Company and the terms of this Agreement, except as required by Law. The Seller Representative shall deliver to Buyer a draft copy of each such Tax Return so as to afford Buyer a reasonable opportunity to review and comment on such Tax Return (not less than thirty (30) days prior to the applicable filing deadline in the case of any income tax return), and Sellers shall make all changes reasonably requested by Buyer and based on Law. Sellers shall not amend any such Tax Returns without the written consent of Buyer, not to be unreasonably withheld.
- 8.4 Responsibility for Filing Straddle Period Tax Returns. Buyer shall at its cost prepare or cause to be prepared, and timely file or cause to be timely filed, all Tax Returns of the Company with respect to any Straddle Period. Any such Tax Return shall be prepared in a manner consistent with prior practices of the Company and the terms of this Agreement, except as required by Law. Buyer shall deliver to the Seller Representative a

102596957.7 0059466-00008

draft copy of each such Tax Return so as to afford the Seller Representative a reasonable opportunity to review and comment on such Tax Return (not less than thirty (30) days prior to the applicable filing deadline in the case of any income tax return), and make all changes reasonably requested by the Seller Representative. Buyer shall not amend any such Tax Returns without the written consent of the Seller Representative, not to be unreasonably withheld.

8.5 Transfer Taxes. All Transfer Taxes incurred in connection with the transactions contemplated by this Agreement shall be paid by the Party incurring such Taxes under applicable Law when due. The responsible Party shall, at its own expense, file all necessary Tax Returns and other documentation with respect to all such Transfer Taxes. The filing Party shall deliver to the other Parties a draft copy of each such Tax Return so as to afford the other Parties a reasonable opportunity to review and comment on such Tax Return, and make all changes reasonably requested by the other Parties.

#### 8.6 Cooperation on Tax Matters.

- (a) Each Seller, the Company and Buyer shall cooperate as and to the extent reasonably requested by any Party hereto in connection with the preparation and filing of Tax Returns as provided herein and any audit, litigation or other proceeding with respect to Taxes related to the Company. Such cooperation shall include the provision of records and information which are reasonably relevant to any such Tax Return, audit, litigation or other proceeding and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder.
- (b) Buyer and Sellers further agree, upon request, to use reasonable efforts to obtain any certificate or other document from any Governmental Authority or any other Person as may be necessary to mitigate, reduce or eliminate any Tax that could be imposed.
- (c) Buyer and Sellers further agree, upon request, to provide any other Party with all information that such Party may be required to report pursuant to Section 6043 of the Code and all Treasury Regulations promulgated thereunder.
- 8.7 <u>Tax Sharing Agreements</u>. All tax-sharing agreements or similar agreements with respect to or involving the Company shall be terminated as of the Closing and, after the Closing, the Company shall not be bound thereby or have any liability thereunder.
- 8.8 Withholding. Buyer shall have the right to withhold any amount payable to Sellers under this Agreement as is required by Law and, to the extent such amounts are paid over to the applicable taxing authority and the Seller Representative is given notice of such withholding, all such amount shall be treated as paid to Sellers under this Agreement.

## ARTICLE IX INDEMNIFICATION

#### 9.1 Survival.

- (a) Survival of Representations and Warranties. The representations and warranties in this Agreement and the schedules and exhibits attached hereto or in any writing delivered by any Party to any of the other Parties in connection with this Agreement shall survive the Closing as follows:
  - the representations and warranties in <u>Section 5.12</u> (Tax Matters), <u>Section 5.17</u> (Employees), <u>Section 5.19</u> (Employee Benefit Plans) and <u>Section 5.22</u> (Environmental Matters) shall terminate on the later of (A) the five (5) year anniversary of the Effective Date or (B) the date when the applicable statutes of limitations with respect to the Liabilities in question expire (giving effect to any extensions or waivers thereof), plus sixty (60) days;
  - the representations and warranties in <u>Section 4.1</u> (Power of Sellers), <u>Section 4.2</u> (Authorization; No Breach), <u>Section 4.3</u> (Brokerage), <u>Section 4.4</u> (Title to Membership Interests), <u>Section 4.5</u> (Investment), <u>Section 5.1</u> (Organization and Power), <u>Section 5.2</u> (Authorization; No Breach), <u>Section 5.3</u> (Capitalization; Subsidiaries), the first sentence of <u>Section 5.11</u> (Assets), <u>Section 5.16</u> (Brokerage), <u>Section 6.1</u> (Organization and Corporate Power), <u>Section 6.2</u> (Authorization), and <u>Section 6.4</u> (Brokerage) (each of the foregoing, a "<u>Fundamental Representation</u>") shall not terminate; and
  - (iii) all other representations and warranties in this Agreement and the schedules attached hereto or in any writing delivered by any Party to any other Party in connection with this Agreement shall terminate eighteen (18) months following the Closing;

provided, that any representation or warranty in respect of which indemnity may be sought under Section 9.2, and the indemnity with respect thereto, shall survive the time at which it would otherwise terminate pursuant to this Section 9.1 if notice of the inaccuracy or breach or potential inaccuracy or breach thereof giving rise to such right or alleged right of indemnity shall have been given to the Party against whom such indemnity may be sought prior to such time. The representations and warranties in this Agreement and the schedules attached hereto or in any writing delivered by any Party to any other Party in connection with this Agreement shall survive for the periods set forth in this Section 9.1 and shall in no event be affected by any investigation, inquiry or examination made for or on behalf of any Party, or the knowledge of any Party's officers, directors, shareholders, employees or agents, as applicable, or the acceptance by any Party of any certificate or opinion hereunder. There shall be no limitation on the time

- within which notice of a claim based on fraud or Willful Breach on the part of any Party may be made.
- (b) Survival of Covenants. Each covenant or agreement in this Agreement and the schedules attached hereto or in any agreement, certificate, instrument or writing delivered by any Party to any other Party in connection with this Agreement shall survive the Closing and be enforceable until such covenant or agreement has been fully performed.

#### 9.2 Indemnification.

- (a) Indemnification for Benefit of Buyer. Each Seller shall indemnify, defend and save and hold each of the Buyer Indemnified Parties harmless against, and pay on behalf of or reimburse each of the Buyer Indemnified Parties as and when incurred for any Losses which any of the Buyer Indemnified Parties may suffer, sustain or become subject to, as a result of, in connection with, relating or incidental to, or by virtue of:
  - (i) any breach or inaccuracy of any representation or warranty of any Seller under this Agreement or any of the schedules attached hereto (including the Disclosure Schedule), or in any of the agreements, certificates or other instruments or documents furnished to Buyer by any Seller or the Company or their Affiliates pursuant to this Agreement;
  - (ii) any nonfulfillment, nonperformance or breach of any covenant or agreement by any Seller or their Affiliates (including, prior to the Closing, the Company) under this Agreement, any of the schedules attached hereto (including, without limitation, the Disclosure Schedule), or in any of the agreements, certificates or other instruments or documents furnished to Buyer by any Seller, the Company or their Affiliates pursuant to this Agreement;
  - (iii) to the extent the following items are not included in the calculation of the final Closing Cash Amount, without duplication: (A) any Closing Indebtedness; (B) any Non-Ordinary Course Liabilities; and (C) any Excluded Liabilities; and
  - (iv) those facts, circumstances, events, occurrences or matters set forth on Exhibit 9.2(a)(iv).

Each Seller shall be jointly and severally liable to the Buyer Indemnified Parties for any and all Losses subject to indemnification pursuant to this <u>Section 9.2(a)</u> up to such Seller's Pro Rata Share of such Losses; provided, however, that a Seller shall not be liable to Buyer or the Buyer Indemnified Parties to the extent (but only to the extent) of those Losses subject to indemnification pursuant to this <u>Section 9.2(a)</u> that result solely and directly from (A) breach or inaccuracy of any representation or warranty made by and with respect to another Seller pursuant to

on the part of such other Seller pursuant to <u>Sections 7.3, 7.4, 7.5</u> or <u>7.7</u>; provided, further, that each Seller shall be fully liable to Buyer and the Buyer Indemnified Parties to the extent of those of those Losses subject to indemnification pursuant to this <u>Section 9.2(a)</u> that arise in connection with, relating or incidental to, or by virtue of (X) breach or inaccuracy of any representation or warranty made by and with respect to such Seller pursuant to <u>ARTICLE IV</u>, or (Y) nonfulfillment or breach by such Seller of any covenant on the part of such Seller pursuant to <u>Sections 7.3, 7.4, 7.5</u> or <u>7.7</u>, or (Z) fraud, intentional misrepresentation, intentional breach, willful misconduct or criminal activity on the part of such Seller.

- (b) Indemnification for Benefit of Sellers. Buyer shall indemnify the Seller Indemnified Parties and save and hold each of them harmless against, and pay on behalf of or reimburse each of the Seller Indemnified Parties as and when incurred for any Losses which any of the Seller Indemnified Parties may suffer, sustain or become subject to, as a result of, in connection with, relating or incidental to, or by virtue of:
  - any breach of any representation or warranty of Buyer under this Agreement, any of the schedules attached hereto, or in any of the agreements, certificates or other instruments or documents furnished to Sellers by Buyer pursuant to this Agreement; and
  - (ii) any nonfulfillment or breach of any covenant or agreement by Buyer under this Agreement or any of the schedules attached hereto or in any of the agreements, certificates or other instruments or documents furnished to Sellers by Buyer pursuant to this Agreement.
- 9.3 <u>Materiality</u>. For purposes of determining any breach or inaccuracy of a representation or warranty and the calculation of Losses arising in connection therewith under this <u>ARTICLE IX</u>, each representation and warranty contained in this Agreement or in any of the agreements, certificates or other instruments or documents furnished pursuant to this Agreement, shall be read and interpreted without regard and without giving effect to any qualification as to materiality, Material Adverse Effect or other similar qualification contained in or otherwise applicable to such representation or warranty.

#### 9.4 Indemnification Procedures.

(a) Manner of Payment. Any indemnification of the Buyer Indemnified Parties or Seller Indemnified Parties pursuant to Section 9.2 shall be effected by wire transfer of immediately available funds from Sellers or Buyer, as the case may be, to an account designated by Buyer or the Seller Representative, as the case may be, within ten (10) days after the determination thereof. Any such indemnification payments shall include interest at the Applicable Rate, calculated on the basis of the actual number of days elapsed over 360, from the date any such Loss is suffered or sustained to the date of payment. Any amounts that Buyer reasonably believes are owing from any Seller pursuant to Section 9.2(a) may, at the option of Buyer, be paid from the Escrow Amount in accordance with the Escrow Agreement, or as a set off against any Adjustment Amount (as set forth in Section 2.3(b)(vi)), if any, or other amount then due and owing from Buyer to Sellers, or as a set off against any Buyer Stock delivered as part of the Closing Stock Consideration or otherwise held by Sellers, and shall otherwise be paid directly by Sellers, jointly and severally, in accordance with the terms of this Section 9.4. All indemnification payments under this Section 9.4 shall be treated for Tax purposes as adjustments to the Purchase Price, unless otherwise required by applicable Law.

- (b) Defense of Third Party Claims. Any Party making a claim for indemnification under Section 9.2 (an "Indemnitee") shall notify the indemnifying Party (or Parties) (each, an "Indemnitor") of the claim in writing promptly after receiving written notice of any action, lawsuit, proceeding, investigation or other claim against it by a third party that is not an Affiliate of the Indemnitee (each, a "Third Party Claim"), describing the Third Party Claim, the basis therefor and the amount thereof (if known and quantifiable); provided, that the failure to so notify any Indemnitor shall not relieve the Indemnitor of its obligations hereunder except to the extent (and only to the extent) that such failure shall have caused the damages for which the Indemnitor is obligated to be greater than such damages would have been had the Indemnitee given the Indemnitor prompt notice of such Third Party Claim hereunder. Any Indemnitor shall be entitled to participate in the defense of such action, lawsuit, proceeding, investigation or other claim giving rise to an Indemnitee's claim for indemnification at such Indemnitor's sole cost and expense; provided, that the Indemnitee shall at all times retain control of the defense of such Third Party Claim.
- 9.5 Waiver. Each Seller hereby agrees that such Seller and its Affiliates shall not make any claim for indemnification hereunder against the Company by reason of the fact that such Seller, its Affiliates, or one or more of their respective employees, agents or representatives was a member, manager, director, officer, employee or agent of the Company or was serving at the request of the Company as a partner, trustee, manager, director, officer, employee or agent of another entity (whether such claim is for judgments, damages, penalties, fines, costs, amounts paid in settlement, losses, expenses or otherwise) with respect to any claim, action, lawsuit, proceeding, investigation brought by any Buyer Indemnified Party against such Seller, and each Seller hereby acknowledges and agrees that such Seller shall have no claims or right to contribution or indemnity from the Company with respect to any amounts paid by such Seller pursuant to Section 9.2.

#### 9.6 Limitations on Indemnity.

(a) <u>Basket</u>. No claim for indemnification shall be made by the Buyer Indemnified Parties pursuant to <u>Section 9.2(a)(i)</u> or by the Seller Indemnified Parties pursuant to <u>Section 9.2(b)(i)</u> with respect to an individual item of Losses subject to indemnification unless and to the extent that the aggregate of all such Losses incurred by the Buyer Indemnified Parties or the Seller Indemnified Parties.

102596957.7 0059466-00008

- respectively, under this Agreement shall be in excess of \$50,000, whereupon Sellers and Buyer, as the case may be, shall be liable for all such Losses from the first dollar.
- (b) <u>Cap</u>. The maximum aggregate liability of Buyer for Losses pursuant to <u>Section 9.2(b)(i)</u>, on the one hand, or Sellers for Losses pursuant to <u>Section 9.2(a)(i)</u>, on the other hand, under this Agreement shall be \$1,880,000.
- (c) Exceptions. Notwithstanding the foregoing, the limitations set forth in this Section 9.6 shall not apply to Losses: (i) arising as a result of, in connection with, relating or incidental to or by virtue of fraud, intentional misrepresentation, intentional breach, willful misconduct, criminal activity or the breach or inaccuracy of any Fundamental Representation; or (ii) the breach or inaccuracy of the representations and warranties set forth in the second sentence of Section 5.6 or in Sections 5.12, 5.17, 5.19 or 5.22; or (iii) subject to indemnification pursuant to Sections 9.2(a)(ii), 9.2(a)(iii), 9.2(a)(iv) and 9.2(b)(ii).
- 9.7 Exclusive Remedy. The Parties acknowledge and agree that their sole and exclusive remedy with respect to any claim relating to the subject matter of this Agreement shall be pursuant to the indemnification provisions of this ARTICLE IX, except for claims involving: (a) fraud, intentional misrepresentation or Willful Breach of any covenant, in which case the Parties shall have all rights and remedies available under this Agreement, at law, in equity or by statute (which remedies shall be cumulative, and not alternative or exclusive); and/or (b) nonfulfillment or breach of any covenant or agreement under this Agreement, or in any agreement or other agreement, certificate or instrument furnished pursuant to this Agreement, in which case the Parties shall be entitled to seek equitable relief (including an injunctive relief or specific performance), without the requirement of posting a bond or other security, in addition to all other rights and remedies available in accordance with this Agreement. Sellers acknowledge and agree that the covenants of each Seller set forth in this Agreement are unique and valuable to Buyer and that Buyer will not have a sufficient remedy at law, and shall be entitled to seek specific enforcement of such covenants, in the event of any actual or threatened nonperformance or breach of such covenants by any Seller.

## ARTICLE X MISCELLANEOUS

10.1 Amendment and Waiver. This Agreement may be amended, and any provision of this Agreement may be waived; provided, that: (a) any such amendment or waiver shall be binding upon Sellers only if set forth in a writing executed by the Seller Representative and referring specifically to the provision alleged to have been amended or waived; and (b) any such amendment or waiver shall be binding upon Buyer only if set forth in a writing executed by Buyer and referring specifically to the provision alleged to have been amended or waived. No course of dealing between or among the Parties shall be deemed effective to modify, amend or discharge any part of this Agreement or any rights or obligations of any Party under or by reason of this Agreement, and a waiver of any

provision by any Party on one occasion shall not be deemed to be a waiver of the same or any other breach on a future occasion.

Notices. All notices, demands and other communications to be given or delivered under or by reason of the provisions of this Agreement shall be in writing and, shall be deemed to have been given as of the earlier of: (a) when personally delivered, or sent by facsimile (with hard copy to follow); (b) one (1) day following dispatch by reputable overnight express courier, charges prepaid; (c) three (3) days following mailing by U.S. certified or registered mail, postage prepaid and return receipt requested; or (d) when sent by email. Unless another address is specified in by notice given pursuant to this Section 10.2, all notices, demands and communications to the Parties shall be sent to the addresses indicated below:

Notices to any or all of Sellers or the Seller Representative:

Dustin B. Baker 10774 Township Road 506 Shreve, Ohio 44676

Email: dbaker@midflowservices.com

Brant Baker 389 Marshall Heights Drive Wexford, PA 15090

Email: bbaker@midflowservices.com

with a copy (which shall not constitute notice to Sellers or the Seller Representative) to:

Critchfield, Critchfield & Johnston, Ltd. 138 East Jackson Street Millersburg, Ohio 44654 Attn: Garrett M. Roach Facsimile No.:(330)674-4469 Email: roach@ccj.com

#### Notices to Buyer:

Profire Energy, Inc. 321 South 1250 West Lindon, UT 84042

Attn: Todd Fugal Facsimile No.: (801) 785-5545

Email: tfugal@profireenergy.com

#### with a copy (which shall not constitute notice to Buyer) to:

Stoel Rives LLP 201 S. Main Street, Suite 1100 Salt Lake City, UT 84111

Attn: Samuel P. Gardiner Facsimile No.: (801) 578-6999

Email: sam.gardiner@stoel.com

#### 10.3 Assignment.

- (a) This Agreement and all of the provisions hereof shall be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns, except that neither this Agreement nor any of the rights, interests or obligations hereunder may be assigned or delegated by any Seller without the prior written consent of Buyer. Buyer may assign its rights, interests or obligations under this Agreement only to the extent set forth in Section 10.3(b) below.
- (b) Buyer may assign in whole or in part its rights and obligations pursuant to this Agreement (including the right to purchase the Membership Interests) and all other agreements, documents and instruments executed and/or delivered in connection herewith to one or more of its Affiliates, and Buyer may, in its sole discretion, direct Sellers to convey the Membership Interests, in whole or in part, to one or more of Buyer's Affiliates. Buyer may assign this Agreement, all other agreements, documents and instruments executed and/or delivered in connection herewith, and its rights and obligations hereunder and thereunder, in connection with a merger or consolidation involving Buyer or in connection with a sale of stock (or other ownership interests) or assets of Buyer or other disposition of all or any portion of Buyer's business. Buyer may assign any or all of its rights pursuant to this Agreement, including its rights to indemnification, and all other agreements, documents and instruments executed and/or delivered in connection herewith, to any of its lender(s) as collateral security.
- 10.4 Severability. Whenever possible, each provision of this Agreement shall be interpreted in such manner as to be effective and valid under applicable Law, but if any provision of this Agreement or the application of any such provision to any Person or circumstance shall be held to be prohibited by or invalid, illegal or unenforceable under applicable Law in any respect by a court of competent jurisdiction, such provision shall be ineffective only to the extent of such prohibition, invalidity, illegality or unenforceability, without invalidating the remainder of such provision or the remaining provisions of this Agreement. Furthermore, in lieu of such prohibited, illegal, invalid or unenforceable provision, there shall be added automatically as a part of this Agreement a permissible, legal, valid and enforceable provision as similar in terms to such illegal, invalid, or unenforceable provision as is possible, permissible and consistent with the intentions of the Parties as set forth in this Agreement.

- 10.5 Interpretation. The headings and captions used in this Agreement and the table of contents to this Agreement are for reference purposes only, and shall not affect in any way the meaning or interpretation of this Agreement. Any capitalized term used in any schedule or exhibit attached hereto, and not otherwise defined therein, shall have the meaning set forth in this Agreement. The use of the word "including" herein shall mean "including without limitation." The Parties hereto intend that each representation, warranty and covenant contained herein shall have independent significance. If any Party has breached any representation, warranty or covenant contained herein in any respect, the fact that there exists another representation, warranty or covenant relating to the same subject matter (regardless of the relative levels of specificity) which such Party has not breached shall not detract from or mitigate the fact that such Party is in breach of the first representation, warranty or covenant.
- 10.6 Entire Agreement. This Agreement and the agreements and documents referred to herein contain the entire agreement and understanding between the Parties hereto with respect to the subject matter hereof and supersede all prior agreements and understandings, whether written or oral, relating to such subject matter in any way.
- 10.7 Counterparts. This Agreement may be executed in one or more counterparts, each of which will be deemed an original, and all of which together will be deemed to be one and the same instrument binding upon the Parties, notwithstanding the fact that all of the Parties are not signatory to the original or the same counterpart. Counterparts to this Agreement and the other agreements, certificates and instruments delivered pursuant hereto may be signed, transmitted and delivered electronically, by email in portable document format (.pdf) or by other electronic means which create or preserve a retrievable image of an original signature, and such counterparts shall be deemed originals for all purposes.
- 10.8 Governing Law. THIS AGREEMENT AND ALL QUESTIONS OR DISPUTES CONCERNING THE CONSTRUCTION, VALIDITY, INTERPRETATION AND ENFORCEABILITY OF THIS AGREEMENT AND THE SCHEDULES ATTACHED HERETO, AND THE PERFORMANCE OF THE OBLIGATIONS IMPOSED BY THIS AGREEMENT, SHALL BE GOVERNED BY THE INTERNAL LAWS OF THE STATE OF UTAH, WITHOUT GIVING EFFECT TO ANY CHOICE OF LAW OR CONFLICT OF LAW RULES OR PROVISIONS.
- 10.9 Venue. The Parties hereby agree that any action, suit or proceeding arising out of or related to this Agreement shall be conducted only in the courts of the United States or the State of Utah located in Salt Lake County, Utah. Each Party hereby voluntarily, knowingly and irrevocably consents and submits to the exclusive personal jurisdiction of and venue of such courts located in Salt Lake County, Utah. Each Party agrees to accept service of any summons, complaint or other initial pleading made in the manner provided for the giving of notices in Section 10.2. Nothing contained in this Section 10.9 shall be construed to limit or preclude a Party from bringing any action in any court of competent jurisdiction for injunctive or other provisional relief to compel any other Party to comply with its obligations under this Agreement or any other agreement between or among the Parties at any time.

54

102596957.7 0059466-00008

- 10.10 No Strict Construction. Notwithstanding the fact that this Agreement may have been drafted and prepared by or on behalf of one of the Parties, the Parties confirm that they and their respective counsel have reviewed, negotiated and adopted this Agreement as the joint agreement and understanding of the Parties, and the language used in this Agreement shall be deemed to be the language chosen by the Parties hereto to express their mutual intent, and no rule of strict construction shall be applied against any Person.
- 10.11 No Third-Party Beneficiaries. This Agreement is for the sole benefit of the Parties hereto and their permitted assigns and nothing herein expressed or implied, shall give or be construed to give any Person, other than the Parties hereto and such permitted assigns, any legal or equitable rights hereunder.

#### 10.12 Seller Representative.

- (a) <u>Designation</u>. Dustin B. Baker (the "<u>Seller Representative</u>") is hereby designated by Sellers to serve as the representative of Sellers with respect to the matters expressly set forth in this Agreement to be performed by the Seller Representative.
- (b) Authority. Each Seller, by the execution of this Agreement, hereby knowingly, voluntarily and irrevocably appoints the Seller Representative as the agent, proxy and attorney-in-fact for such Seller for all purposes of this Agreement, with the full power and authority on such Seller's behalf to: (i) consummate the transactions contemplated herein; (ii) execute and deliver to Buyer any confirmatory assignments with respect to the Membership Interests and such further instruments of transfer as Buyer shall reasonably request; (iii) pay such Seller's expenses incurred in connection with the negotiation and performance of this Agreement (whether incurred on or after the date hereof); (iv) disburse any funds received hereunder to such Seller; (v) execute and deliver any amendment or waiver hereto; (vi) take all other actions to be taken by or on behalf of such Seller in connection herewith; (vii) negotiate, settle, compromise and otherwise handle adjustments to the Closing Consideration as set forth in Section 2.3(b) and all claims for indemnification made by any of the Buyer Indemnified Parties pursuant to Section 9.2 hereof; and (viii) do each and every act and exercise any and all rights which such Seller is permitted or required to do or exercise under this Agreement. Each Seller acknowledges and agrees that such agency and proxy are coupled with an interest, are therefore irrevocable without the consent of the Seller Representative and shall survive the death, incapacity, bankruptcy, dissolution or liquidation of such Seller.
- (c) <u>Exculpation</u>. Neither the Seller Representative nor any agent employed by it shall incur any Liability to any Seller by virtue of the failure or refusal of the Seller Representative for any reason to consummate the transactions contemplated hereby or relating to the performance of its other duties hereunder, except for actions or omissions constituting fraud or bad faith.

10.13 <u>Disclosure Schedule.</u> Nothing disclosed in the Disclosure Schedule shall qualify or create an exception to a representation or warranty made in this Agreement unless such disclosure identifies the exception with reasonable particularity and describes the relevant facts in reasonable detail. Without limiting the generality of the foregoing, the mere listing (or inclusion of a copy) of a Contract, document or other item shall not qualify or disclose an exception to a representation or warranty made in this Agreement, unless the representation or warranty has to do with the existence of the document or other item itself. No qualifications or exceptions to any representations or warranties disclosed on one Section of the Disclosure Schedule shall apply to any other representations or warranties made in this Agreement unless (and then only to the extent that) such qualification or exception is disclosed as provided herein on each such other applicable Section of the Disclosure Schedule or the applicability of such disclosure is readily apparent as a qualification or an exception to such other representation or warranty. The Disclosure Schedule is hereby incorporated in and made a part of this Agreement as if set forth in full herein.

IN WITNESS WHEREOF, the Parties hereto have caused this Membership Interest Purchase Agreement to be duly executed as of the Effective Date.

<b>Buyer:</b> PROFI	RE ENERGY, INC.
By: Name: Title: _	Brenton W. Hato
Sellers	;
Dustin	Baker, individually
Brant I	Baker, individually
Seller	Representative:
<u> </u>	B. Baker, Seller Representative

[Signature Page to Membership Interest Purchase Agreement]

IN WITNESS WHEREOF, the Parties hereto have caused this Membership Interest Purchase Agreement to be duly executed as of the Effective Date.

By:		
Name:		101 1000
Title:	THE CONTRACTOR OF THE CONTRACT	
Sellers:		
/ Dustin B	aker, individually	3/
The state of the s	ker, individually	
Seller R	epresentative:	

[Signature Page to Membership Interest Purchase Agreement]

#### EXHIBIT 31.1

### CERTIFICATION OF PRINCIPAL EXECUTIVE OFFICER Pursuant to Rule 13a-14(a) of the Securities Exchange Act of 1934

- I, Brenton W. Hatch, certify that:
  - 1. I have reviewed this quarterly report on Form 10-Q of Profire Energy, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a) Designed such disclosure controls and procedures or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures as of the end of the period covered by this report based on such evaluation; and
  - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

control over financial	,	management or othe	r employees	wno nave a	significant for	e in the registra	int s internal
Date	:: November 6, 2019		-	/s/ Brenton W. Brenton W. Ha			

Chief Executive Officer

#### EXHIBIT 31.2

#### CERTIFICATION OF PRINCIPAL FINANCIAL OFFICER Pursuant to Rule 13a-14(a) of the Securities Exchange Act of 1934

#### I, Ryan W. Oviatt, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Profire Energy, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a) Designed such disclosure controls and procedures or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures as of the end of the period covered by this report based on such evaluation; and
  - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.				
Date:	November 6, 2019	Ву:	/s/ Ryan W. Oviatt Ryan W. Oviatt Chief Financial Officer	

# CERTIFICATION OF PRINCIPAL EXECUTIVE OFFICER PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with this quarterly report on Form 10-Q of Profire Energy, Inc. (the "Company") for the period ended September 30, 2019, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Brenton W. Hatch, Chief Executive Officer of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: November 6, 2019

By: /s/ Brenton W. Hatch
Brenton W. Hatch
Chief Executive Officer

# CERTIFICATION OF PRINCIPAL FINANCIAL OFFICER PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with this quarterly report on Form 10-Q of Profire Energy, Inc. (the "Company") for the period ended September 30, 2019, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Ryan W. Oviatt, Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: November 6, 2019

By: /s/ Ryan W. Oviatt

Ryan W. Oviatt

Chief Financial Officer